

<mark>_</mark>

Water Resources Research

RESEARCH ARTICLE

10.1029/2019WR025388

Special Section:

Advancing Process Representation in Hydrologic Models: Integrating New Concepts, Knowledge, and Data

Key Points:

- A promising integrated hydrological modeling approach is proposed to assess the future effects of climate change and groundwater pumping
- Gaining streams will be more impacted by the climate change scenarios and pumping variants than losing streams
- Groundwater overutilization turns out to be the major reason for the extreme groundwater storage drawdown in the aquifer of the Gharehsoo River Basin, Iran

Correspondence to:

M. Taie Semiromi, majid.taie@zalf.de

Citation:

Taie Semiromi, M., & Koch, M. (2020). How do gaining and losing streams react to the combined effects of climate change and pumping in the Gharehsoo River Basin, Iran? *Water Resources Research*, *56*, e2019WR025388. https:// doi.org/10.1029/2019WR025388

Received 17 APR 2019 Accepted 9 MAR 2020 Accepted article online 18 MAR 2020

©2020. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

How Do Gaining and Losing Streams React to the Combined Effects of Climate Change and Pumping in the Gharehsoo River Basin, Iran?

M. Taie Semiromi^{1,2} D and M. Koch¹

¹Department of Geohydraulics and Engineering Hydrology, University of Kassel, Kassel, Germany, ²Now at Working Group "Lowland Hydrology and Water Management", Leibniz Centre for Agricultural Landscape Research (ZALF), Müncheberg, Germany

Abstract Projections of potential impacts of climate change and groundwater abstraction on gaining and losing streams, particularly in ephemeral river basins exhibiting sporadic and intricate flux exchanges, have remained largely unexplored. To fill this gap, we propose a promising modeling scheme based on the new fully integrated hydrological model SWAT-MODFLOW-NWT, calibrated and validated for 1978-2012, to quantify the intertwined surface-groundwater interactions under a conjuncture of three climatic emission scenarios (RCP 2.6, 4.5 and 8.5) and two groundwater pumping variants: "pumping" (extending current groundwater utilization into the future) and "nonpumping" (assuming a complete cease of pumping in the future). By forcing the integrated model with future downscaled climate predictors of CanESM2 under the aforementioned RCPs for three time slices up to year 2100, projections of various water resources components for the Gharehsoo River Basin (GRB), in northwestern Iran were made. Results demonstrate that because of a general decrease of future precipitation, though with ups and downs across the total projection period, most of the surface and -subsurface budget quantities and fluxes are substantially affected. In particular, future groundwater discharge (baseflow) to the gaining streams will be more influenced by the "pumping" variant (increasing and decreasing for "nonpumping" and "pumping", respectively) than the concentrated groundwater recharge from the losing streams (decreasing and increasing for "nonpumping" and "pumping", respectively). Future water yield and groundwater storage will also diminish and, surprisingly, this cannot be alleviated by future "nonpumping", indicating the groundwater overutilization is the compelling reason for the future water scarcity in the GRB, rather than climate change alone.

Plain Language Summary Understanding climate change impacts on surface-groundwater interactions, accommodating nearly all of water resources components of a watershed, is of paramount importance in devising effective strategies to alleviate the adverse impacts of climate change. This is particularly important in semiarid regions where the surface water scarcity has markedly increased the dependency on groundwater resources which, in turn, has highly affected the interaction between these two. We have provided an innovative scheme to assess and project climate change impacts on water resources of a basin and, in particularly, on gaining and losing streams at different temporal and spatial scales. In addition, the net impacts of climate change impacts on the quantity of the water resources, compared with the groundwater overutilization, can be quantified using the proposed methodological approach. Our findings importantly highlight that the groundwater overutilization is the major reason, rather than climate change impacts, for extreme groundwater storage depletion in the Gharehsoo River Basin, in northwestern Iran. As a result, groundwater discharge to the river network will diminish tremendously under the changing climate projected for the region.

1. Introduction

According to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (IPCC), "climate change over the 21st century is projected to reduce renewable surface water and groundwater resources in most dry subtropical regions (robust evidence, high agreement), intensifying competition for water among sectors (limited evidence, medium agreement)" (IPCC, 2013). In fact, water resources have been affected by global change all over the world in a wide verity of unpredictable ways (Green et al., 2011; Mani et al., 2016). Potential impacts of climate change on surface hydrological processes have been discussed in further details, but as emphasized by Holman, 2006, Green et al., 2007, Green et al., 2011, Bovolo et al., 2009, Nkhonjera & Dinka, 2017, Nistor & Mindrescu, 2019, not many studies have drawn attention to impacts posed on the subsurface hydrological processes.

Surface water resources are directly affected by climate change impacts owing to changes in the major long-term hydroclimate variables such as air temperature, precipitation, and evapotranspiration. However, groundwater resources are inextricably interwoven with the changing climate variables, thereby posing formidable difficulty in quantifying their potential impacts (P. Goderniaux et al., 2009; Jyrkama & Sykes, 2007; Sanderson & Curtis, 2016). Groundwater resources response to climate change occures directly through the flux exchange with surface water bodies, including lakes, wetlands, and river networks (e.g., Dwire et al., 2017; Goderniaux et al., 2011; P. Goderniaux et al., 2009; Scibek et al., 2007; Taviani & Henriksen, 2015; van Engelenburg et al., 2018; Yidana et al., 2019). Groundwater reaction to climate change effects takes place indirectly via the recharge processes (e.g., Bovolo et al., 2009; Crosbie et al., 2010; Erturk et al., 2014; Gemitzi et al., 2017; Ghazavi & Ebrahimi, 2019; Goodarzi et al., 2016; Jyrkama & Sykes, 2007; Kahsay et al., 2018; Nassery & Salami, 2016; Riasati et al., 2012; Scibek & Allen, 2006; Shrestha et al., 2016; Smerdon, 2017). In addition, dependency on the groundwater consumption can be increased in the wake of less availability of surface water under a changing climate (e.g., Goodarzi et al., 2019). Furthermore, the interactions between climate change and groundwater may become even increasingly complex due to the anthropogenic activities (Ashwell et al., 2017), such as land-use change (LUC) (Mohamed et al., 2017; Taylor et al., 2013) and groundwater abstractions (W. Liu et al., 2020).

The fluctuations of the flow in gaining and losing streams in a river network connected hydraulically to the aquifers are highly associated with the recharge variabilities (Taie Semiromi & Koch, 2019). In addition, since climate change produces major effects on the recharge rate, providing a reliable estimation of the later necessitates simulation of the atmospheric, surface, and subsurface hydrologic processes using a fully integrated mechanistic hydrological model (P. Goderniaux et al., 2009). Recharge amounts can be estimated using a wide variety of techniques ranging from simple to complex approaches. These are conventionally classified as water budget-based methods (e.g., Tilahun & Merkel, 2009), groundwater modeling techniques (e.g., J. W. Chen et al., 2016), surface-water data-based techniques (e.g., Meyboom, 1961), physical methods: unsaturated zone (e.g., Min et al., 2015), physical methods: saturated zone (e.g., Varni et al., 2013), chemical tracer approaches (e.g., Wu et al., 2016), and heat tracer methods (e.g., Taniguchi & Sharma, 1993). Notwithstanding their differences, none of the aforementioned approaches can address flux and feedback exchanges taking place between the surface and groundwater domains. Due to the fact that the meteoric infiltration (recharge) is reliant upon precipitation and evapotranspiration over the land surface, evapotranspiration within the vadose and saturated zone, and ultimately river-aquifer interactions, this feedback becomes an inherent component of the water cycle. Thus, to accurately quantify these processes, applying a fully integrated mechanistic hydrological model is inevitable. This is owing to the fact that the involved processes, including the flux feedback, can be simulated both simultaneously and separately. Taking into consideration either one part of the hydrological system, namely, surface water or groundwater, will yield unrealistic, inaccurate, and potentially unusable results in the context of climate change impact assessments (P. Goderniaux et al., 2009).

Despite the big advantages of using fully integrated mechanistic models for the assessment of climate change impacts on the subsurface hydrological components, most studies have enjoyed employing simplified schemes thus far for such a purpose. For instance, Z. H. Chen et al. (2002) investigated climate change impacts on a Canadian aquifer using a simple linear transfer function, which connected the groundwater head fluctuations to the variabilities of precipitation and temperature. Obviously, these types of simplifications between recharge and groundwater head introduce considerable uncertainty. To reduce the latter, created as a result of an independently assessment of surface and groundwater domains under a changing climate, fully coupled/integrated physics-based and spatially distributed hydrological models have recently come to the fore (Taviani & Henriksen, 2015). Thus far, a number of fully coupled surface-subsurface hydrological models have been developed such as HydroGeoSphere (Brunner & Simmons, 2012), CATHY (Camporese et al., 2010), ParFlow (Maxwell et al., 2007), InHM (VanderKwaak & Loague, 2001), OGS (Kolditz et al., 2012), and SWAT-MODFLOW-NWT (Bailey et al., 2016). In comparison with other integrated physic-based hydrological models, SWAT-MODFLOW-NWT takes advantages of two models, namely, SWAT and MODFLOW-NWT, where either is the most commonly used model for surface and subsurface

hydrological modeling, respectively. SWAT-MODFLOW-NWT has been developed in a manner to produce a complementary ability for having the advantages of either of the two models, while overcoming the potential disadvantages when they are used individually (Taie Semiromi & Koch, 2019). Conventionally, surface and subsurface hydrological models have mostly been coupled with the aim of quantifying the interactions between a river network and its adjoining groundwater system (Bejranonda et al., 2007; Kalbus et al., 2006).

Employing physically based hydrological modeling approaches taking into consideration both surface and subsurface processes in order to project the climate change impacts on surface-groundwater interactions has not been well documented as of yet. To that respect, a few studies have investigated the interconnection between a groundwater system and a river network under a changing climate (Goderniaux et al., 2011, 2015; P. Goderniaux et al., 2009; Saha et al., 2017; Scibek et al., 2007). Moreover, a few works have explored the response of groundwater-dependent ecosystems, including wetlands, lakes, and riparian areas to climate change scenarios (Dwire et al., 2017; Havril et al., 2017; Klove et al., 2014; Taviani & Henriksen, 2015; van Engelenburg et al., 2018).

Although in previous studies climate change impacts on one of the compartments of surfacewater or groundwater has been projected, the climate change impacts upon the surface-groundwater interactions, as reflected in gaining (effluent) and losing (influent) streams, groundwater storage, and the ultimate product of the water cycle in a basin, namely, the so-called water yield, have not been assessed till date. In this regard, Scibek et al. (2007) studied groundwater heads and streamflow hydrographs under climate change and groundwater pumping scenarios. However, they did not assess the impacts of climate change and pumping scenarios on the gaining and losing streams, where the flux exchanges between groundwater domain and river network (Alvarez-Villa et al., 2014; Baalousha, 2012). P. Goderniaux et al. (2009), Goderniaux et al. (2011), and Goderniaux et al. (2015) projected climate change impacts on surface and groundwater components using the HydroGeoSphere (HGS) model, a finite-element coupled surface-subsurface flow model (Brunner & Simmons, 2012). However, the central focus of these studies was to provide groundwater head and streamflow fluctuations under the Special Report on Emission Scenarios (SRES) A2 emission (medium-high) scenario. Thus, a spatiotemporal analysis of the climate change scenario on the surface-groundwater interactions represented by the gaining/losing streams in relation to their adjoining groundwater domain was not conducted. Notwithstanding the previous studies, Saha et al. (2017) explored the groundwater discharge/baseflow generated from gaining reaches under climate change scenarios A2 and B1 of SRES using a physically based and distributed model, that is, Gridded Surface Subsurface Hydrologic Analysis (GSSHA). Although the baseflow, supplied by the gaining streams, was projected for 2020-2040 for different temporal scales, the spatial variability of this water component across the Kiskatinaw River Watershed in British Columbia, Canada, was not evaluated. In addition, the potential impacts of climate change on the concentrated groundwater recharge, produced by the losing streams/reaches, towards the shallow groundwater were not taken into consideration.

Given the mentioned research gaps in the assessment of climate change impacts on surface-groundwater interactions, in the current study, we introduce a new hybrid modeling scheme for a comprehensive assessment of climate change effects on different tangible components of the hydrological cycle, in particular, the surface-groundwater flux exchange. To that end, we will benefit from a recently developed and well-tested fully coupled hydrological model, namely, SWAT-MODFLOW-NWT (Bailey et al., 2016; Taie Semiromi & Koch, 2019), in which the land surface processes, in-stream, vadose zone, and subsurface hydrological processes are efficiently and consistently coupled. In the light of using this coupled model, highly resolved spatiotemporal outputs of baseflow/groundwater discharge resulting from the gaining streams, as the conventional output of most hydrological models, in response to various climate change scenarios, can be projected. Additionally, the same projection can be made for the losing streams where much of the groundwater recharge takes place. In addition to the surface-groundwater exchange processes, this model enables one to assess a wide range of other water resources components of a catchment such as groundwater storage, total streamflow, and water yield in response to different stressors such as climate change effects.

There is an urgent necessity for climate impact and hydrological studies in the Gharehsoo River Basin (GRB), located in northwestern Iran where an intermittent/seasonal drainage network dictates a complex surface-groundwater interaction mechanism. First, inasmuch as the study area is located in a mountainous region, it is expected to be more sensitive to global climate change (Huber et al., 2005; Shi & Durran, 2014), as

is the case for many mountainous areas of the world (e.g., the Andes and the Himalayas) (Hu et al., 2013). Second, as reported by IPCC (2013), in many mid-latitude and subtropical dry regions such as the GRB, the mean precipitation will most likely decrease in the future. This will have enormous adverse impacts on the water availability in the GRB. For these reasons, projections of spatiotemporal impacts of climate on a wide range of water resources components, particularly those supplied by gaining and losing streams, under different emission scenarios are very important for proposing effective adaptation and mitigation strategies in this region.

Third, the fragile balance between water supply and demand has recently been lost in the GRB, and supposedly, the ongoing climate change impacts will further worsen the situation. To that respect, Ardabil Regional Water Authority (2013) has expressed a need for climate change impacts assessment studies on the various water resources components of this basin. This is especially required in the GRB where the low flows, as a highly transient flow, are dependent on the sporadic and intricate interactions between the shallow groundwater and losing/gaining streams. Indeed, these low flows can supply a major fraction of agricultural, industrial, and domestic consumptions for this basin, especially, during dry periods, that is, summer.

Thus, the objective of this study is to advance our understanding of gaining and losing streams under climate change impacts by employing a fully integrated hydrological model, SWAT-MODFLOW-NWT, as the latter can resolve several simplifications considered in previously mentioned studies. Above all, in this study, a recently developed state-of-the-art hybrid model employing artificial neural networks (ANNs), discrete wavelet transform (DWT), and quantile mapping (QM) is used for the downscaling of daily precipitation (Taie Semiromi & Koch, 2017). Moreover, this study makes use of the Second Generation Canadian Earth System Model (CanESM2) developed by the Canadian Centre for Climate Modeling and Analysis (CCCma). Climate projections are available under three representative concentration pathways (RCPs), namely, RCP 2.6, RCP 4.5, and RCP 8.5. Moreover, to distinguish man-made effects from climate change impacts on surface-groundwater interactions, the projections will be made under two further variants, namely, (1) "nonpumping" variant (undisturbed system: assuming a complete cease of pumping in the future) and (2) "pumping" variant (disturbed system: extending the current groundwater utilization into the future).

2. Application Area

2.1. Gharehsoo River Basin

The Gharehsoo River Basin (GRB), situated in the northwest of Iran (Figure 1a), covers an area of 4,193 km². It is located between latitudes 37°46′ to 38°36′N and longitudes 47°46′ to 48°42′E (Javan et al., 2015). Recognized as a semiarid region, dry and wet seasons of the GRB span from May to October and November to April, respectively. Considering the historical daily records obtained from synoptic meteorology stations of the Iran Meteorological Organization and climatology stations of the Iran Ministry of Energy in the study area, the hydroclimate variables, including daily minimum and maximum temperature and precipitation of 12 weather stations (Figure 1b), distributed over the GRB for the observed period 1978–2005 were collected. The statistics and geographical characteristics of the selected weather stations are listed in Table 1 and 2. On average, the annual precipitation across the GRB is around 305.90 mm (Table 1); however, their average varies spatially from 267.11 mm for the plain area/aquifer of the basin to 357.36 mm in the mountainous zone (Figure 1b). The precipitation falls mainly in the cold months, that is, November to April. According to the Köppen climate classification, the prevailing climate in this region is "cold and semiarid" (BSk).

2.2. Ardabil Aquifer

The Ardabil plain/aquifer (38°30′ to 38°27′N and 47°55′ to 48°20′E), located within the GRB, covers an area of 1,073 km² (Figures 1a–1d). The aquifer is encircled by two high mountains, Alborz and Sabalan. On average, the areal annual precipitation over the aquifer is about 270 mm, and the prevailing climate identifies May and August as the wettest and driest months of the year, respectively (Ardabil Regional Water Authority, 2013; Nourani et al., 2015). With an average temperature of 9 °C, the Ardabil aquifer is known for being one of the coldest regions in Iran, as characterized by an average of 130 freezing days in a year





Figure 1. Geographical location of the Gharehsoo River Basin (GRB) on Iran's map (a), spatial variation of the precipitation averaged for the historic period and location of the meteorological stations considered in this study as well as the position of the aquifer/Ardabil plain on the basin (b), digital elevation model (DEM) of the GRB (c), and land-use/land-cover map of the basin along with the drainage network pattern over the basin and the modeled area/Ardabil aquifer (d).



Statistics of the Precipitation Recorded for the Reference Period 1978-2005 at 12 Weather Stations Distributed Over the GRB

Climate station	Latitude (°N)	Longitude (°E)	Elevation (m)	Minimum (mm)	Maximum (mm)	Average (mm)	Standard deviation (mm)
Abi Biglou	38.28	48.56	1,348.00	89.38	429.80	262.43	79.17
Abi Biglou	38.28	48.56	1,348.00	89.38	429.80	262.43	79.17
Ardebil	38.22	48.33	1,335.20	189.34	534.11	298.57	83.14
Gilandeh	38.31	48.37	1,341.00	163.00	361.00	241.72	60.19
Hir	38.08	48.51	1,594.00	173.03	499.50	294.66	91.80
Kouzeh	38.12	48.37	1,402.00	120.00	357.00	235.30	64.78
Lay	38.12	47.91	2,038.00	194.61	580.16	356.21	95.15
Namin	38.41	48.47	1,405.00	269.50	439.23	335.95	55.38
Nir	38.04	48.02	1,623.00	226.00	714.50	368.11	120.67
Pole Almas	38.16	48.20	1,480.00	152.00	492.00	296.41	76.60
Samian	38.37	48.25	1,286.00	157.61	492.50	239.09	67.23
Shams Abad	38.00	48.25	1,534.00	170.00	642.00	317.77	125.77
Yamchi Olya	38.04	48.50	1,594.00	209.79	586.42	333.19	95.25

(Vousoughi et al., 2013) (Table 2). Demographically, according to the national census conducted in 2011, the population of the plain is around 564,000 that inhabits two big cities, including the provincial city Ardabil, and 88 villages (Statistical Center of Iran, 2011). Farming is the major professional occupation in this plain. As a result of massive expansion of Ardabil city, developing irrigated agriculture, and industrial developments, an increasing pressure on the aquifer has been exerted since the 1980s. The annual water consumption of the drinking, industrial, and agricultural sectors amounts to 26, 4, and 177 million cubic meters per year (MCM/yr), respectively; out of which 89% is obtained from groundwater (Ardabil Regional Water Authority (2013). The groundwater supply emanates from 2,622 active pumping wells, 36 qanats, and 77 springs operating across the Ardabil aquifer (K. Kord & Asghari Moghaddam, 2014).

2.3. Hydrogeology and Boundary Condition

Quaternary alluvial deposits resulting from the erosion of the Alborz and Sabalan mountains form the Ardabil plain (Figure 2a) (K. Kord & Asghari Moghaddam, 2014). In the light of geophysical surveys, pumping test analysis, and drilling logs undertaken for the Ardabil aquifer which can provide useful information on the structure of aquifer systems (Jellalia et al., 2015), the Ardabil aquifer is found to be composed of gravel, sand, and a small fraction of clay. The aquifer can be divided into two layers, including a major uppermost unconfined aquifer with a thickness varying from 20 to 40 m, and a tiny confined layer within its central part. However, since these two layers are laterally hydraulically connected, isolating them from one another cannot be a plausible option, especially, when a surface-groundwater interaction investigation is aimed. Most of the pumping wells have been drilled through the unconfined aquifer, with only a small fraction penetrating into the lower layer which lies between a depth of 110 and 220 m (M. Kord et al., 2013). The thickness of the alluvium gradually decreases from east to west. The greatest alluvial thickness of the aquifer,

The Same as Table 1, But for the Min and Max Temperature									
Climate st	tation	Latitude (°N)	Longitude (°E)	Elevation (m)	Minimum (°C)	Maximum (°C)	Average (°C)	Standard deviation (°C)	
Ardebil	T _{Max}	38.22	48.33	1,335.20	-12.00	39.80	15.43	9.49	
	T _{Min}				-33.80	21.00	2.77	8.18	
Airport	T _{Max}	38.33	48.42	1,314.30	-11.18	38.34	15.00	9.09	
	T _{Min}				-36.89	19.83	0.95	8.51	
Namin	T _{Max}	38.42	48.49	1,480.20	-8.23	47.93	15.14	10.32	
	T _{Min}				-30.96	18.17	1.85	7.33	
Nir	T _{Max}	38.04	48.03	1,593.00	-13.60	39.65	14.60	9.75	
	T _{Min}				-33.28	16.15	-0.27	7.37	
Sareyn	T _{Max}	38.15	48.08	1,658.30	-14.57	37.00	11.34	9.00	
	T _{Min}				-28.66	23.52	4.60	7.80	

Note. Although some of the stations' names are similar in Tables 1 and 2, they have different geographical positions as they are operated by two different authorities, that is, Iran Meteorological Organization (IRIMO) and Iran Ministry of Energy.

Table 2





Figure 2. Ardabil aquifer geological formation map (a), and groundwater flow direction map with aquifer boundaries to be used in the groundwater flow model (b) (following Taie Semiromi and Koch (2019)).

reaching to 220 m and representing a thick saturated zone, composed mainly of coarse grain sediments, is located in the east and southwest parts of the plain/aquifer. The groundwater heads measured by the piezometric stations, used for monitoring the qualitative and quantitative characteristics of the groundwater in the Ardabil aquifer, are employed to draw the groundwater contour map, as well as to delineate the flow direction (Figure 2b). The groundwater heads vary from 1,529 to 1,280 MSL (meters above mean sea level) and follow the prevailing topography (Figure 1c); therefore, the cardinal direction of the groundwater flow is towards the northwest of the plain (Figure 2b). To assign the hydraulic aquifer properties including specific yield (*Sy*) and transmissivity (*T*), pumping induced drawdowns at 40 observation wells/piezometers, distributed evenly across the aquifer, were used. Following Todd and Mays (2005), pumping wells having no observation wells were employed for the estimation of the transmissivity (*T*). The aquifer's *T* was found to vary widely from 50 to 2,200 m²/day, and its *Sy* from 0.021 to 0.14 (M. Kord et al., 2013).

The boundary conditions of the numerical groundwater flow model (MODFLOW-NWT, Niswonger et al., 2011) were assigned by respecting the geological formations encompassing the aquifer as well as the observed groundwater contour pattern (see Figure 2). To that end, flow/head boundary conditions were specified along most sections of the Ardabil aquifer boundary by means of MODFLOW's FHB-package (Flow Head Boundary), thereby attributing in a general way no-flow, inflow, and outflow boundaries, as illustrated in Figure 2b. Groundwater inflow and outflow into the model domain were calculated using Darcy's law and specified along the northwest boundary of the aquifer where not only all surface water is drained out of the main outlet of the basin (Samian station) (Figure 1b), but where also groundwater outflow occurs. Some sections of the aquifer boundary, that are in contact with impervious formations, were specified as no-flow, although according to the groundwater contours (see Figure 2b), some inflow may enter the aquifer. Due to the fact that some installed piezometers are quite close to the aquifer boundary, time-variant specified heads (MODFLOW's CHD-package) and recognized as Dirichlet conditions were assigned at places where historical observed groundwater heads were available, for example, P18 (Figure 2b).

3. Methodology

3.1. Constructing, Calibrating, and Validating the SWAT-MODFLOW-NWT Coupled Model

The coupled hydrological model employed for this study is the so-called SWAT-MODFLOW-NWT. The model is based on integrating MODFLOW-NWT (Newton-Raphson formulation of the Modular Three-Dimensional Finite-Difference Groundwater Flow model of the eminent MODFLOW model of McDonald et al. (1988)) (Niswonger et al., 2011), with SWAT (Soil and Water Assessment Tool; J.G Arnold et al., 1998), as developed by Bailey et al. (2016). For the purpose of coupling, the SWAT and MODFLOW-NWT models were independently constructed, calibrated, and validated against discharge measured at five hydrometric stations (Figure 1b) and groundwater heads recorded at 68 observation wells (Figure 2b), respectively. Subsequently, a recalibration was made for the hydraulic conductance of the stream network, where the SWAT and MODFLOW-NWT models' flow domains essentially meet. The finally coupled SWAT-MODFLOW-NWT model was constructed, calibrated, and validated for the 1978-2012 time period wherefore the first 3 years were considered as a spin-up period to allow the water cycle to reach an equilibrium and thus to ensure an initial state (Daggupati et al., 2015). Therefore, the complete study period, that is, 1978-2012, was split into three sub-periods, 1978-1980, 1988-2012, and 1981-1987 for spin-up, calibration, and validation, respectively. The magnitude and direction of the flow between a gaining/losing stream and its adjacent groundwater domain are determined according to the MODFLOW's River package. To that respect, SWAT calculates the stream depth based on Manning's equation, assuming a trapezoidal shape for the reaches/streams, and the SWAT-estimated volume of flow (Neitsch et al., 2011). Then, by adding the computed stream depth to the stream bed elevation extracted from the digital elevation model (DEM), the effective stream stage is calculated to be used in the SWAT-MODFLOW-NWT model (Bailey et al., 2016).

To parameterize the river-aquifer interaction of the coupled model and thus providing a high resolution of the spatial variability of this interaction, the river network was incised in the MODFLOW-grid cells using GIS operations. As a result, 2,581 reaches were created. The magnitude and direction of flow, characterized in gaining and losing reaches, were determined using an adaption of Darcy's law (equation 1).





$$QRIV_{n} = CRIV_{n} (HRIV_{n} - h_{i,j,k})$$
(1)

where $QRIV_n$ (L^3T^{-1}) quntifies the volumetric flow between a reach and its adjacent aquifer (McDonald, Harbaugh,, & Geological Survey (U.S.), 1988) and $CRIV_n$ (L^2T^{-1}) is the hydraulic conductance of the reach's bed material as calculated by equation 2.

$$CRIV_n = \frac{K_n L_n W_n}{M_n}$$
(2)

where K_n is the hydraulic conductivity of the reach's bed material with thikness M_n , L_n is the length of the reach through the grid cell (i, j, k), and W_n represents the reach's width.

In integrated hydrological modeling, the river-aquifer interaction is normally parameterized using the river conductance $CRIV_n$, as characterized via a head-dependent flux (Cauchy-type) boundary condition (Cousquer et al., 2017; Furman, 2008; Pascal Goderniaux et al., 2009). As a lump parameter, it cannot be measured at the field scale due to its combination of different parameters/processes. Accordingly, it is calibrated using a history matching approach (Ebel et al., 2009; McDonald et al., 1988; Rushton, 2007), wherefore the simulated variables such as groundwater heads and streamflow discharge are assessed against their observed values. Nevertheless, the calibrated values should be further check out by the soft knowledge of the modeler (Anderson et al., 2015).

In the present study, the hydraulic conductance of the reaches $CRIV_n$ was calibrated using a trial-and-error procedure by minimizing the discrepancy between observed and SWAT-MODFLOW-NWT- simulated groundwater/streamflow discharge. Ultimately, the water yield as the final product of the river-aquifer flux exchange, surface runoff, lateral flow, and tile flow was calculated as:

Water Yield =
$$SURQ + LATQ + TILEQ + GWQ - SWGW$$
 (3)

where SURQ denotes the surface runoff/direct runoff, LATQ the interflow/lateral flow, TILEQ the tile flow, GWQ the groundwater discharge/baseflow generated from the gaining reaches, and SWGW indicates the concentrated recharge towards the adjacent aquifer produced in the losing streams (Park & Bailey, 2017). It should be noted that in the application area, that is, the GRB, there is no tile drainage system. Therefore, the term TILEQ in equation 3 was "zero" in this study.

The details of the individual calibrations and validations of each model (SWAT, MODFLOW-NWT) and of the coupled one are described by Taie Semiromi and Koch (2019). In principle, this integrated hydrological model affords a dynamic variable exchange between SWAT and MODFLOW-NWT. To that respect, first, the deep percolation/recharge estimated by SWAT is passed to MODFLOW-NWT; second, the river heads calculated by SWAT are fed to the River package (RIV) of MODFLOW-NWT; and third, the computed ground-water discharge/baseflow of the gaining reaches is delivered to SWAT and, similarly, the concentrated groundwater recharge of the losing reaches computed by the River package (RIV) of MODFLOW-NWT is subtracted from the total streamflow (direct surface runoff + baseflow + lateral flow). A flowchart of the exchanged variables of the proposed coupled modeling as described is sketched in Figure 3a; with the details of the various methodological steps provided by Bailey et al. (2016) and Taie Semiromi and Koch (2019).

3.2. Downscaling of Hydroclimate Variables

In the present study, daily minimum and maximum temperature at five weather stations distributed across the GRB (Figure 1b) were downscaled using the well-known Statistical DownScaling Model (SDSM) (Wilby et al., 2002) which has found innumerable applications in climate change studies of the last two decades (see Wilby & Dawson, 2013, for a review). The daily historical data records available at these stations representing the reference period fall within the 1978–2005 time period. Climate projections were available for RCP (Representative Concentration Pathways) 2.6, RCP 4.5, and RCP 8.5 up to year 2100. These RCPs represent the range of GHG emissions reported in the wider literature well; they include a stringent mitigation scenario (RCP 2.6), two intermediate scenarios (RCP 4.5 and RCP 6.0), and one scenario with very high GHG emissions (RCP 8.5). After screening the predictors, based on the strongest correlations between various climate predictors (e.g., near-surface temperature, and mean sea level pressure) and observed weather variables (min and max temperatures), depending on the weather station between two and three predictors





Figure 3. Methodology proposed for the projection of the various water resources components of the GRB using (a) the coupled model SWAT-MODFLOW-NWT, forced by (b) daily downscaled min and max temperature via SDSM model and (c) daily downscaled precipitation using the DWT-ANN-QM model.

were selected to build the transfer (regression) statistical model. Observations of two third of the reference period, 1978–1996, were used to calibrate the statistical model, and the remainder of the data, that is, 1997–2005, was set aside to validate the model. Subsequently, the calibrated model was applied to the corresponding predictors of the Second Generation Canadian Earth System Model (CanESM2) model under the named RCPs to project min and max temperatures for the future.

The daily precipitation was downscaled for 12 weather stations (Figure 1b) using a state-of-the-art hybrid statistical downscaling model that takes advantages of artificial neural networks (ANNs), discrete wavelet transform (DWT), and quantile mapping (QM) (Taie Semiromi & Koch, 2017). Using a stepwise regression procedure statistical relationships between the National Centers for Environmental Prediction (NCEP)-predictors and the daily precipitation at each station for the reference period 1978–2005 were established and thus the most suitable predictor(s) were screened, similar to above. Daily predictors of CanESM2 under RCPs 2.6, 4.5, and 8.5 were then employed to force the developed hybrid model for the future projection period 2006–2100. The coupled wavelet and neural network (DWT-ANN) model employs various subseries components of the DWT-multiresolution analysis of Level 4 of the selected climate predictors at a weather station. These decomposed predictors were used as inputs to train (calibrate) the model with respect to the observed precipitation (predictand) as output. Afterwards, the DWT-ANN-simulated daily precipitation was bias-corrected using the QM algorithm.

3.3. Projection of the Effects of Climate Change Scenarios and Pumping Variants on the Various Water Resources Compartments

Once the daily maximum and minimum temperature and precipitation data were successfully downscaled, the coupled hydrological model, that is, SWAT-MODFLOW-NWT, was forced by the downscaled hydroclimate variables under six scenarios/variants (three \times two) including three RCPs 2.6, 4.5, and 8.5, with each of them applied to the coupled model under two groundwater utilization variants, namely, "pumping" and "nonpumping". By doing so, one can differentiate the net impacts of climate change from that of

groundwater overutilization. Given the fact that the Ardabil aquifer has been strictly regulated to not be further exhausted by new pumping wells (Ardabil Regional Water Authority, 2013), the number of pumping wells and therefore the amount of groundwater extracted for the future projection period for the "pumping" variant were assumed to be as much as the last year of the hydrologic model simulation (2012), namely 2,622 wells, extracting a total of 160 MCM per year (K. Kord & Asghari Moghaddam, 2014). This pumping rate corresponds to an equivalent water height across the total area of the Ardabil aquifer of about 150 mm/yr, which is nearly 15 times higher than the total diffuse surface-to-groundwater recharge of 10 mm/yr during the historic period, demonstrating clearly the extreme overutilization of this aquifer. For the "nonpumping" variant on the other hand, all these wells were shut off in the coupled hydrological model during the future simulation periods, in order to be able to separate out the hydrological effects due to groundwater pumping from those due to climate change. A similar approach was taken by Scibek et al. (2007) in their study of the impacts of climate change and pumping on surface-groundwater interaction in a Canadian watershed. Although the reference simulation period for the SWAT-MODFLOW-NWT model is 1978-2012, conventionally, the climate variables of the GCMs for the reference period are available to only up to year 2005. Accordingly, the projections are made from year 2006 onwards (e.g., 2006-2100). Since we already had the observed variables for the calibration and validation of the hydrological model by 2012, we forced the coupled model by the downscaled variables (min and mam temperature and precipitation) from 2013 onwards, that is, 2013-2100, although the hydroclimate variables were projected for the future period 2006-2100.

Finally, the spatiotemporal variability of the various water resources components of the GRB was extracted for three future time slices: 2013–2040 (near future), 2041–2070 (middle future), and 2071–2099 (distant future) and under the six aforementioned climate/pumping condition combinations. These water resources components include meteoric infiltration (diffuse recharge), baseflow emanating from the gaining streams, concentrated groundwater recharge contributed by the losing streams, baseflow index (amount of baseflow divided by amount of total streamflow), groundwater storage, and water yield as calculated by equation 3. It should be noted that since the coupled hydrological model produces the output variables at a daily temporal scale, they were then aggregated and averaged to produce monthly outputs.

4. Results and Discussion

4.1. Downscaling of Min and Max Temperature

The results of the SDSM-downscaled min temperature show (see Figure 4) that, in comparison with that of the reference period (1978–2005), a notable increase in min temperature, particularly, during winter (e.g., January and February), will occur under all three emission scenarios (RCP 2.6, 4.5, and 8.5), wherefore (and not unexpectedly) this increase is more pronounced for RCP 8.5 and the distant future (2071–2100). Nevertheless, a minor decrease in min temperature is to be expected in September for all stations except for Sareyn, for which the coldest days records on a national scale have been reported several times (Ardabil Regional Water Authority, 2013). On an annual scale, the lowest increases in the annual average of min temperature are projected for Namin station, ranging from 0.65 °C under RCP 2.6 and time slice 2006–2040 to 2.5 °C under RCP 8.5 and time slice 2071–2100, while the highest increases in min temperature are projected for Sareyn station, ranging from 0.93 °C under RCP 2.6 and time slice 2006–2040 to 3.59 °C under RCP 8.5 and time slice 2071–2100.

Similarly, as illustrated by Figure 5, the downscaled future max temperature will be considerably raised at all stations and months, except for April. Similar to the downscaled min temperature, the max temperature will be dramatically increased when going from RCP 2.6 to 8.5 and from the near future, 2006–2040, to the distant future, 2071–2100. On average, the lowest increases of the annual average of max temperature are predicted for Sareyn station, ranging from 2.87 °C under RCP 2.6 and time slice 2006–2040 to 8.54 °C under RCP 8.5 and time slice 2071–2100, while the highest increases in max temperature are observed for Nir station, ranging from 3.65 °C under RCP 2.6 and time slice 2006–2040 to 9.7 °C under RCP 8.5 and time slice 2071–2100. On this matter, Javan et al. (2015) projected the mean temperature in the GRB for the distant future (2071–2100) by employing the PRECIS (Providing REgional Climates for Impacts Studies) climate model and considering B2 as a SRES (Special Report on Emissions Scenarios) scenario. Their findings revealed that the mean temperature will be raised 2–4 °C in winter, 2–5 °C in spring, 3–5 °C in summer,





Figure 4. Monthly projected minimum temperature under RCP 2.6, 4.5, and 8.5 over three time slices 2006–2040, 2041–2070, and 2071–2100 with that of the reference period (1978–2005) at five weather stations.

and 2–4 °C in autumn. Moreover, they found that the minimum and maximum temperature will occur in months January and July, respectively. Overall, their results are fairly consistent with the projections made in the present study for the distant future (2071–2100) under RCP 4.5.

4.2. Downscaling of Precipitation

The daily precipitation projected by the DWT-ANN-QM model for the three future time slices 2006–2040, 2041–2070, and 2071–2100 under the three emission scenarios RCPs 2.6, 4.5, and 8.5, are illustrated in Figure 6. These projections manifest that, compared with the reference period (1978–2005), the future



Figure 5. Same as Figure 4, but for maximum temperature.





Figure 6. Monthly averages of the projected precipitation at the 12 weather stations under RCPs 2.6, 4.5, and 8.5 for the three time slices 2006–2040, 2041–2070, and 2071–2100, in comparison with the reference period (1978–2005).

Table 3Average of the Projected Annual Precipitation for the Reference and the Three Future Periods Under the Three RCPs										
Weather station	Reference period (mm)	RCP2.6 (mm)			RCP4.5 (mm)			RCP8.5 (mm)		
		2006-	2041-	2071-	2006-	2041-	2071-	2006-	2041-	2071-
	1978-2005	2040	2070	2100	2040	2070	2100	2040	2070	2100
Abi Biglou	262	194	227	206	198	221	190	209	230	270
Ardebil	298	249	277	255	249	270	249	267	282	299
Gilandeh	241	198	213	196	190	196	186	206	191	204
Hir	295	272	287	278	274	279	271	273	277	290
Kouzeh	235	188	228	191	183	210	190	172	211	243
Lay	356	313	364	314	310	358	317	227	256	317
Namin	336	318	354	335	324	336	324	228	236	260
Nir	368	332	399*	356	334	366	354	279	319	379*

Note. "*" indicates the projected precipitation exceeding the average of the annual precipitation for the reference period (1978–2005).

Pole Almas

Shams Abad

Yamchi Olya

Samian

precipitation will decrease under all three scenarios, particularly for the wet season, that is, from January to May for the climate stations Shams Abad, Namin, while less reductions at station Ardebil under all three RCPs, and even a precipitation increase is to be expected during the dry period, that is, June to September at some stations, such as Hir and Shams Abad. This projected rise in precipitation in the future wet seasons over the GRB has also been corroborated by Javan et al. (2015) under the SRES scenario B2 as their findings show that the precipitation will be increased for the future period (2071-2100) for the months January, February, March, September, and December. Moreover, our results indicate further that the reduction in the monthly precipitation under the extreme RCP 8.5 is surprisingly less than that projected under RCP 4.5. This peculiar behavior can be associated with a higher perceptible water supply of the atmosphere resulting from a higher temperature increase under RCP 8.5 than RCP 4.5 (Huang et al., 2014). Overall, the precipitation will decrease in the near future (2013-2040), then it will rise in the middle future (2041-2070), and again decrease in the distant future (2071-2100). In Table 3, the estimated average of the annual precipitation for the reference period (1978-2005) and three future periods (2006-2040, 2041-2070, and 2071–2100) under RCPs 2.6, 4.5, and 8.5 is listed. The numbers in the table corroborate Figure 6; that is, for most of the weather stations, the average annual precipitation will decrease for the three future time periods, relative to the historic one. As indicated by a "*" in Table 3, some minor increases are also expected for stations Lay, Nir, Namin, Samian, Shams Abad, and Yamchi Olya for the time span 2041-2070 under RCP 2.6; for station Lay for the same period under RCP 4.5; and for stations Abi Biglou, Ardebil, Kouzeh, Nir, and Yamchi Olya for the time slice 2071-2100 under RCP 8.5.

A climate study conducted by Terink et al. (2013), using nine GCMs and SRES A1B, 2013 demonstrated that the annual precipitation will decline by 15-20% for the majority of countries across the Middle East and Northern Africa for the two future periods 2020–2030 and 2040–2050, which appears to confirm the reduction witnessed in the present study, namely, a 12% decline in the precipitation over the GRB, on average. Nonetheless, in some studies, by contrast, a rise in the future precipitation has been reported. For instance, Modarres et al. (2018) projected the variability of the maximum daily rainfall across the North and Northwest of Iran for 2020–2049 using six GCMs under the two scenarios A2 and B1. In particular, their results suggest the projected precipitation to vary from -16 to 7% under scenario A2, compared with the reference period (1981–2010). Likewise, depending on the GCM used, the future maximum rainfall may change from -16 to 10% under scenario B1, relative to the reference period.

4.3. Projection of Diffuse Recharge Under the Various Climate Change Scenarios

The SWAT-MODFLOW-NWT-estimated monthly recharge (deep percolation) over the GRB for the reference simulation period (1978–2012), the projected near future (2013–2040), middle future (2040–2070), and distant future (2071–2099) periods under RCPs 2.6, 4.5, and 8.5 are represented in Figure 7. One can notice that the recharge will be increased from months of January to April (mostly winter season) for all three future time periods under RCPs 2.6 and 4.5, while such an increase is only expected for January, February, and April for the middle and distant futures under RCP 8.5. However, in the rest of the months, the recharge will be nearly the same as that of the reference simulation period (e.g., July to October, except for the distant future period 2071–2099) or will decline (e.g., March and May under RCP 8.5). In addition to these future changes of the magnitude of the recharge, Figure 7 shows further that its timing will most likely also be changed, due to an anticipated winter warming (Woldeamlak et al., 2007). Thus, one can notice from Fgure 7 that, whereas the maximum recharge occurs in March during the reference simulation period, it will be retarded and moved to April, and this holds for all three future time periods and the three RCPs.

It is interesting to note that this one-month retardation of the future maximum recharge occurs despite the fact the projected precipitation (see Figure 6), will decrease in April, irrespective of the future time slices and emission scenarios considered. However, as a result of the notable projected increase in the min and max temperature for the future winters (see Figures 4 and 5), and due to the fact that a major fraction of the precipitation is received and stored as snow, particularly in the mountainous region of the basin (Ardabil Regional Water Authority, 2013), an earlier hillslope snowmelt runoff, and correspondingly, earlier recharge may be expected (Hayashi & Farrow, 2014). In this study, in contrast, the maximum recharge is projected to be shifted 1 month later, namely, April. This happens because a major portion of the precipitation, as the main driver of the hydrological cycle (e.g., J. G. Arnold et al., 1993), is evapotranspirated during the winter seasons, owing to the considerable projected rise in the min and max temperature. However, as Figures 4





Figure 7. SWAT-MODFLOW-NWT-estimated monthly diffuse recharge for the historic simulation period (1981–2012) with the projected diffuse groundwater recharge under RCPs 2.6, 4.5, and 8.5 for the three time slices 2013–2040, 2041–2070, and 2071–2099.

and 5 indicate, the temperature will not be raised in the future months of April, so that more water for the diffuse recharge will be available during that month.

4.4. Projection of Surface Runoff Under the Various Climate Change Scenarios and Pumping Variants

The SWAT-MODFLOW-NWT-predicted climate change impacts on direct runoff/surface runoff shown in Figure 8 demonstrate that for all three future periods and all three emission scenarios RCP 2.6, 4.5, and 8.5, this hydrological quantity will considerably decline (Figure 8), compared to that of the reference simulation period (1978-2012) and this hold for both ("pumping" and "nonpumping") variants. As illustrated in Figure 8, this reduction in the surface runoff is more (expectedly) pronounced when going from RCP 2.6 to 8.5 as well as from the near future (2013-2040) to the distant future (2071-2099). However, the pumping variants exert no effect on the magnitude and timing of the surface runoff. Interestingly, similar to the projected recharge above (see Figure 7), the maximum projected surface runoff will shift from March to April, for the same reasons as discussed earlier. Indeed, surface runoff is partitioned from the same driver (effective precipitation) of the hydrological cycle, with the latter predicted to be majorly reduced during the future winter seasons owing to a projected large rise in the min and max temperature during that period. And as a nonsignificant change in the temperature projected for April, less evapotranspiration is expected, so that, in turn, a larger fraction of the precipitation - which was shown to decrease under all emission scenarios and all time slices in April (see Figure 6) – may partition off into the runoff during that month. This peculiar behavior may be ascribed to some counterbalance of the nonsignificant changes in the min and max temperature and the projected reductions in the precipitation. In addition, given the fact that, on average, 24%, 25%, 37%, and 67% of the total precipitation is received as snow at Ardebil, Samian, Nir, and Lay stations, respectively (Ardabil Regional Water Authority, 2013), and owing to an increase in the precipitation





Figure 8. SWAT-MODFLOW-NWT-estimated monthly surface runoff for the historic/simulation period (1981–2012) to the projected surface runoff for the two pumping variants ("nonpumping" (a–c) and "pumping" (d–f)) for the three time-slices, 2013–2040 (a) and (d); 2041–2070 (b) and (e); and 2071–2100 (c) and (f) under the three RCPs 2.6, 4.5 and 8.5.

projected for the winter for most time slices and emission scenarios, the stored snow will start melting in warmer months (e.g., April). Accordingly, this can further increase the occurrence of the maximum surface runoff, witnessed for April. These results are to be contrasted with the statements of the review of Huber et al. (2005) and of the Canadian watershed study of Scibek et al. (2007) who found that an earlier snowmelt and, correspondingly, an advanced hydrograph peak generally arises in mountainous regions in the wake of global climate change. Unlike the findings of this study, Scibek et al. (2007) also discovered that a noticeable increase in winter discharge will occur in future climate scenarios, most likely due to an increase of rain and snowmelt volumes during the winter season under a warmer climate.

4.5. Projection of the Gaining Streams' Reactions to the Various Climatic Scenarios and Pumping Variants

Due to the fact that the coupled hydrological model SWAT-MODFLOW-NWT has greatly enhanced the estimation of the baseflow/groundwater discharge and since the baseflow generally sustains the river flow, particularly during dry periods in semiarid regions like the GRB, an understanding of the impacts of climate change on this component of the total streamflow is of paramount importance. As baseflow is essentially related to groundwater flow, which, in turn, is strongly affected by external stresses, like pumping, the net impact of the latter on baseflow needs to be investigated and separated from that of climate change. Thus, in the following, the baseflow is analyzed for the two variants of SWAT-MODFLOW-NWT model, that is, "pumping" and "nonpumping." The results, shown in Figures 9a–9c, indicate that for the "nonpumping" model variant, for all future time slices and all of emission scenarios the baseflow will be substantially increased, irrespective of the reduction and increase obtained earlier for the precipitation and the min and max temperature, respectively. However, the same is not true under the "pumping" model variant, since in that case the baseflow will markedly decrease under all emission scenarios and time slices, except for the "near future" when such a baseflow decline cannot be detected. Such a great sensitivity of the baseflow to





Figure 9. Similar to Figure 8, but for the SWAT-MODFLOW-NWT-estimated monthly groundwater discharge (baseflow).

the two pumping variants (and much less to the emission scenario) can be expounded by two important factors: (1) Since the modeled area/Ardabil aquifer is a plain (see Figure 1c) where the groundwater level is close to the land surface, even a slight change of pumping (and less so of climatic) stress can exert a big influence on the groundwater heads (Melo & Wendland, 2017); (2) the groundwater draft from the Ardabil aquifer has dramatically expanded from 35 MCM in 1978 to 160 MCM in 2012, (corresponding to a water height of 150 mm/yr, that is, about 15 times the present-day total diffuse annual groundwater recharge of 10 mm/yr), thus reversing the current groundwater utilization into the "nonpumping" variant (describing an undisturbed system in which the utilization is satisfied by natural springs only) which will lead to a rise of the groundwater heads and, as a result, to an increased discharge contribution from the groundwater towards the gaining streams, albeit a reduction (by-12% on average) in the precipitation and an increase in the min and max temperature.

In Figure 10, the monthly time series of the historic and projected baseflow under the three emission and the two pumping variants are provided, in addition. First of all, one can notice from the three panels of the different RCPs (2.6, 4.5, 8.5) that the differences in the baseflow time series are minor, which indicates that the impacts of the emission scenarios on the hydrological quantity are more or less also negligible. However, there are systematic differences between the two pumping variants. For the "nonpumping" variant, one can notice a discontinuous jump of the baseflow in year 2013, showing the immediate reaction of the system, once the pumping stress is shut-off. From that time onwards, there is monotonous increase of the baseflow, up to about year 2030, and a steady decline thereafter until the end of the future projection period (year 2100), demonstrating the "kick in" of long term climate change. This general course of the baseflow holds also for the "pumping" variant - assuming the current groundwater utilization volume of 160 MCM up to year 2100 -, though its runs with a more or less parallel decrement from its "nonpumping" counterpart. However, under this "pumping" variant, climate change appears to have a measurable effect on the baseflow towards the gainng streams only from about year 2045 onward. All this is a consequence of the fact that the groundwater heads will fall on account of the balance lost between the total groundwater recharge and discharge. Accordingly, owing to Darcy's law implemented in the MODFLOW's RIV package (equation 1), the concentrated groundwater discharge contribution towards the gaining streams will be reduced in the wake of a decline of diffuse groundwater recharge, ensued, in turn, from the reduction of the projected



Figure 10. SWAT-MODFLOW-NWT-simulated historic and projected monthly time series of groundwater discharge (baseflow) to the river network (gaining streams) under the three emission scenarios RCP 2.6 (a), 4.5 (b), and 8.5 (c), and and the two pumping variants ("pumping" and "nonpumping"). Note that the noticeable breaks in 1988 of the historic period result from a a three-year spin up period (an adjustment process for the model to reach an "optimal" state, where internal stores such as groundwater storage move from the estimated initial condition to an "optimal" state (Kim et al., 2018)) considered before running the calibration and validation periods of the coupled model.

precipitation and the rise of the projected min and maximum temperature. Previous studies (e.g., Bovolo et al., 2009; Crosbie et al., 2010; Crosbie et al., 2013; Erturk et al., 2014; Gemitzi et al., 2017; Ghazavi & Ebrahimi, 2019; Goodarzi et al., 2016; Jyrkama & Sykes, 2007; Kahsay et al., 2018; Karian & Dudewicz, 2000; Moeck et al., 2016; Nassery & Salami, 2016; Riasati et al., 2012; Rosenberg et al., 1999; Scibek & Allen, 2006; Shrestha et al., 2016; Smerdon, 2017; Stoll et al., 2011; Thampi & Raneesh, 2012) have broadly linked climate change-induced precipitation changes to spatiotemporal variabilities of groundwater recharge. However, whereas there is an unequivocal rise in temperature due to the global warming, the projected recharge could increase or reduce, depending on how precipitation reacts to climate change effects (Jyrkama & Sykes, 2007).

To characterize in more detail how the baseflow, relative to the surface runoff and lateral flow, will react to the climate change impacts, we calculated the so-called baseflow index BI = baseflow/(surface runoff + lateral flow + baseflow) for the two pumping variants, using these SWAT-MODFLOW-NWT-simulated variables for the historic and projected periods. The results shown in Figure 11 indicate that for both pumping variants, under all three emission scenarios, the future baseflow index will be considerably increased for all months, except for April. Moreover, similar to the previously projected timings of the recharge and surface runoff (1 month delay in the maximum surface runoff and recharge, relative to the reference simulation period) and, unlike the findings of Scibek et al. (2007) who revealed a 1 month earlier peak discharge, the timing of the maximum baseflow index is also projected here to be shifted by one month,





Figure 11. Similar to Figure 8, but for the monthly baseflow index computed from the appropriate SWAT-MODFLOW-NWT-estimated hydrological budget quantities (see definition in the text).

namely from August in the historical period to September in the future projection period (see Figure 4). This shift may be attributed to the projected maximum monthly max temperature witnessed in future August (see Figure 5), but reduced min temperature in future September (see Figure 4). This spectacular rise of the projected baseflow index, in comparison with that of the reference simulation period, can be associated with the noticeable decline of the projected for the direct surface runoff (see Figure 8) which makes up the substantial fraction of the total streamflow in comparison to the baseflow and lateral flow. As such, due to the fact that the surface runoff appears in the denominator of the *BI*-definition index, a smaller surface runoff contribution, projected under the different climate change scenarios here, will deliver a bigger baseflow index. Furthermore, since the baseflow appears in both numerator and denominator of the *BI* formula and, owing to the fact, that the surface runoff and lateral flow are basically not influenced by either of the two pumping variants (see Figure 7), the baseflow index, in turn, will not be affected neither by the latter, and this is exactly what is seen in Figure 11.

4.6. Projection of the Losing Streams' Reactions to the Climatic Scenarios and Pumping Variants

Similar to the gaining streams in the previous section, the reactions of the losing streams to the different climate change scenarios and pumping variants are shown in Figure 12 by the time series of the concentrated groundwater recharge, produced under influent conditions from the losing streams. These graphs indicate that there is a noticeable difference of this quantity between the historic and the projected period, particularly under the emission scenarios RCP 4.5 and 8.5. However, compared with the baseflow of the gaining streams of the previous section, the concentrated recharge of the losing streams here is (1) more than an order of magnitude smaller and (2) less affected by the two pumping variants, that is, "nonpumping" and "pumping. This can be attributed to the fact that the depth to the water table is very shallow in most parts of the Ardabil aquifer, thus allowing for more groundwater contribution towards the river network (effluent conditions), so that influent conditions from losing streams rarely occur (Taie Semiromi & Koch, 2019). Such influent conditions take place mainly across the southern part of the aquifer at locations where a considerable depth to water table has been established due to groundwater overutilization (Figure 2b) (Taie Semiromi & Koch, 2019). Moreover, as discussed earlier, the surface runoff will decrease under the different climate scenarios considered (see Figure 8), which indicate less water will be available to be recharged over





Figure 12. Similar to Figure 10, but for the concentrated groundwater recharge from the river network (losing streams).

the banks and bed of the losing streams to the groundwater. Surprisingly, in contrast to the projected rise of the baseflow under the "nonpumping" variant in the previous section (see Figure 10), Figure 12 shows that the concentrated groundwater recharge from the losing streams will slightly drop in the future under this "nonpumping" variant. This can be ascribed to an overall future rise of the groundwater table, expected in this case. Under this circumstance, as the groundwater heads rise, according to the river-groundwater interaction formulation of Darcy's law (equation 1), the direction of the exchange flow will be reversed, so that some losing streams become gaining ones. In addition to these temporal assessments of the river-groundwater interactions above, the spatial variability of the losing and gaining streams across the study region will be analyzed in the subsequent section.

4.7. Projection of the Spatial Variability of the Gaining and Losing Streams Under the Different Climatic Scenarios and Pumping Variants

One of the distinct advantages of the current SWAT-MODFLOW-NWT coupled hydrological model over previously developed separate surface- and groundwater flow models is that the former is capable to produce high-resolution outputs of the surface-groundwater flux exchanges. According to the flow directions between the 2,581 reaches/streams and the adjacent aquifer, obtained by overlaying the MODFLOW-grids of the groundwater domain and the river network (Taie Semiromi & Koch, 2019) (Figure 1d), we identified 922 and 1,659 gaining and losing streams, respectively. However, the results of the SWAT-MODFLOW-NWT- spatially variable projections shown in Figures 13–15 for the three climate scenarios, RCP 2,6, 4.5 and 8.5, respectively, illustrate that the number of the gaining and losing streams will vary widely in the future, depending on the climatic scenarios and the pumping variants considered. Thus, for RCP 2.6 (Figure 13), one can notice that under the "pumping" variant, the maximum daily





Figure 13. Magnitude and spatial variability of the historic and projected surface-groundwater interactions under the emission scenario RCP 2.6 for the "pumping" and "nonpumping" variants for the three time slices 2013–2040, 2041–2070, and 2071–2099.





Figure 14. Similar to Figure 13, but for RCP 4.5.





Figure 15. Similar to Figure 13, but for RCP 8.5.





Figure 16. Historic and future (for the three time slices as indicated) projected daily total groundwater discharge (top panels) and recharge (bottom panels) to/from the river network under the three emission scenarios (RCPs) for "nonpumping" (left panels) and "pumping" (right panels) variants.

baseflow yielded by the gaining streams (effluent conditions) will be increased from 622.71 m³/day for the historic period (red bars) to 709.29 m^3 /day for the time slice 2041–2070 (the period for which the precipitation is expected to rise) (green bars). Meanwhile, for the "pumping" variant, although there is still an increase of the future maximum daily baseflow, the latter is only about half of that for the "nonpumping" variant (from 622.71 to 673.98), which indicates that pumping reduces clearly the discharge of the groundwater aquifer to the gaining streams. The barplots of Figure 16 condense the numbers of the spatial variability maps above by quantifying the total groundwater discharge/recharge to/from the river network for the three emission scenarios and the two pumping variants. For example, the panel of Figure 16b shows that, in spite of the aforementioned increases of the future maximum daily baseflow, the total amount of daily groundwater discharge (baseflow) across the study region will slightly and continuously drop for the future time slices for the "pumping" variant, irrespective of the RCP. Nonetheless, the same does not hold for the "nonpumping" variant (Figure 16a) where the projected total average daily baseflow will grow from its historic amount of 63416 m³/day to values higher than 70000 m³/day depending on the time slice, irrespective of the RCP. This increase witnessed for the future maximum daily baseflow (groundwater discharge) under the projected climate change in the study region under the "nonpumping" variant scenarios is most likely a result of the more frequent extreme precipitation events that are to be expected in the wake of the global warming predicted over most mid-latitude land masses and across wet tropical regions (a global-scale intensification of heavy precipitation) (IPCC, 2013). This will lead, consequently, to more diffuse surface-to-groundwater recharge, giving rise, in turn, to a shallower groundwater table, thus intensifying the groundwater discharge contribution (baseflow) towards the river network (gaining streams). This is particularly true for the Ardabil aquifer, where the depth to the water table is already very shallow (Ardabil Regional Water Authority, 2013), so that the groundwater residence time of discharge to the streams will be quite short (Modica & Buxton, 1998).

Similar to the gaining streams above, the spatial distribution of the concentrated maximum groundwater recharge (influent condition) of the losing streams taking place over the bed and bank of the streams is illustrated in Figures 13–15 by the blue and brown bars for the historic and future period, respectively. One can see from the size of these bars, that neither the climatic scenario nor the pumping variant exerts a notable influence, unlike of what has been observed for the baseflow of the gaining streams. On the other hand, as shown by the two lower panels of Figure 16, the total daily volume of the concentrated groundwater recharge from the losing streams over the whole aquifer will, however, significantly differ for the three emission scenarios and the two pumping variants. Thus, whereas for the "nonpumping" variant, the future total stream-groundwater recharge is more or less the same as that of the historic period, - though somewhat diminished for the most extreme RPC 8.5 climate scenario -, the opposite is true for the "pumping" variant, where this quantity increases by up to ~20%, namely for the distant future time slice (2071-2100). Nevertheless, in spite of a much higher number of the losing streams, that is, 1,659 rather than 922 for the gaining streams above, the concentrated groundwater recharge from the losing streams is about three orders of magnitude smaller than the groundwater discharge to the gaining streams (baseflow). This is due to the fact that both the hydraulic conductance and the head differences between the losing streams and the adjacent groundwater aquifer are generally rather small (Irvine et al., 2015), namely in the eastern, flatter topography of the GRB. Thus, the maximum daily concentrated groundwater recharge from the losing streams of about 0.4 m³/day, projected under RCP 2.6 and "pumping" for the middle future (2041-2070), takes place mainly across the western part of the aquifer (see Figure 13d).

According to the precipitation rise projected under RCP 2.6 for the time slice 2041–2070 (see Figure 6), a growth in the hillslope surface runoff and, thus, more water to the river network is to be expected. But, on the other hand, the impacts of the higher precipitation on the runoff generation for the scenario and time slice considered may be masked by the unequivocal temperature rise projected for the GRB (Figures 4 and 5). This may be the reason that no noticeable changes can be spotted for the concentrated groundwater recharge from the losing streams, with the projected maximum concentrated groundwater recharge differing only by 0.35 to 0.40 m³/day (see Figures 15a and 13d). Under the "pumping" variant, as an increased groundwater head drawdown is to be expected (Figure 17), the head differences between the losing river's stage and the groundwater table will be increased, so that, the concentrated groundwater recharge will grow. For that reason, the latter will occur under the "pumping" variant. Likewise, the projected daily average of the concentrated groundwater recharge volume over the whole aquifer, in relation to that of the historic period, is also anticipated to increase under the "pumping" variant, regardless of the climatic scenarios considered (Figure 16d). As expected, the concentrated groundwater recharge from the losing streams will fall noticeably under the "nonpumping" variant, and its minimum will take place under RCP 8.5 (Figure 16c). On the other hand, it may be possible that some of the losing streams will convert into gaining streams under the "nonpumping" variant, when a rise in the groundwater heads may occur again. In conclusion of this section, it has been shown clearly that the groundwater contribution, that is, the baseflow towards the gaining streams is to be recognized as the dominant surface-groundwater interaction process for the GRB also in the future, extending the corresponding findings of Taie Semiromi and Koch (2019) for the historic period.

4.8. Projection of Groundwater Storage Under the Various Climatic Scenarios and Pumping Variants

In Figure 17 the climate change impacts on the temporal evolution of the groundwater storage of the aquifer are shown for the different RCPs and the two pumping variants. These time series indicate that under all of three climatic scenarios and for both the "nonpumping" and "pumping" variants, the groundwater storage will monotonously decrease in the future though with a gentler slope than what is observed for the historic period. Surprisingly, even for the "nonpumping" variant, one can see an ongoing reduction of the groundwater storage in the future, and this occurs, despite the fact that groundwater depletion that has been occurring on account of a dramatic overutilization in the recent past (Ardabil Regional Water Authority, 2013; Taie Semiromi & Koch, 2019), should have been reversed to a large extent under this "nonpumping" variant. Thus, by comparing the curves for the two pumping variants in Figure 17, one can conclude that the unsustainable groundwater utilization in the GRB is the determining factor for the observed decrease of the groundwater storage there for the "nonpumping" variant, exacerbated, moreover, by climate change impacts that add even more pressure on the system. Therefore, for this "nonpumping" variant, one can



Figure 17. Historic and projected time series of monthly groundwater storage under the emission scenarios RCP 2.6 (a), 4.5 (b), and 8.5 (c), and the two pumping variants. The noticeable break in 1988 in the historic period results from the discontinuous transition of the three-years pin-up period to the calibration period of the SWAT-MODFLOW-NWT model.

clearly notice that even by assuming an undisturbed system ensured by eliminating the abstraction performed by the 2,622 farming pumping wells extracting around 150 MCM water per year, the initial condition of the first year of the historic simulation period, that is, 1980, cannot be recovered even at the end of the projection period in year 2099. In accordance with the findings of this study, Kirby et al. (2016) affirmed that the future planned irrigation developments in five main regions of Bangladesh can have a larger impact on the water balance components than the projected climate change itself. This is also the case in the present study here, as one cannot distinguish any difference in the groundwater storages projected under the three climatic emission scenarios (RCP 2.6, 4.5 and 8.5) depicted in the three panels of Figure 17. This insensitivity may be explained by the only minor contribution of the diffuse recharge (meteoric infiltration), amounting to 17.8 MCM (equal to only 11% of the total water budget) (Ardabil Regional Water Authority, 2013; Taie Semiromi & Koch, 2019). In addition, in the wake of the earlier projected precipitation decreases (see Figure 6) for the three RCPs considered, the future diffuse surface-to-groundwater recharge will not only become smaller, but also induce more or less equal groundwater storages for these three climate emission scenarios.

4.9. Projection of Water Yield Under the Various Climatic Scenarios and Pumping Variants

Ultimately, we calculated the future total monthly water yield (defined in equation 3) of the GRB as the final product of all surface- and subsurface hydrological contributions, including their interactions, extending the work of Taie Semiromi and Koch (2019) who estimated the water yield in the GRB only for the historic reference period. In this way, the impacts of the different climate change scenarios and of the pumping variants on the future water yield can be assessed. Figure 18 illustrates that under all of three RCPs and for both





Figure 18. Similar to Figure 8, but for the SWAT-MODFLOW-NWT-estimated monthly water yield (equation. 3) for the various combinations of historic and future time periods, RCPs and pumping variants.

pumping ("nonpumping" and "pumping") variants the future monthly water yield is anticipated to fall beneath that of the historic period except for month of April in the time slice 2041-2070 under the (rather benevolent) RCP 2.6, for which an increase in the precipitation has been found earlier for half of the stations in the GRB (see Table 3). Similar to the future projected surface runoff and recharge (see Figures 8 and 9, respectively), the timing of the maximum water yield will, in relation to the historic period, also be shifted 1 month later, owing to the fact that the major fraction of the water yield emanates from the surface runoff which experiences the same fate. It can be further noticed from the various panels of Figure 18, and even more clearly from the time series of the monthly water yield depicted in Figure 19 for the different RCPs / pumping variants that, as expected, the magnitude of the average monthly water yield will generally decline when going from the rather benevolent RCP 2.6 to the most extreme RCP 8.5 climate scenario. Strangely, no future increase of the average water yield under the "nonpumping" variant can be detected, although some sporadic and irregular bursts of the water yield can take place (Figure 19). To explain this kind of peculiar behavior, one should be reminded that, even though an appreciable rise of the baseflow, as one of the contributors to the water yield, has been found earlier under the "nonpumping" variant (see Figures 9 and 10), baseflow generally constitutes only an insignificant fraction of the total streamflow in the GRB. This is because the flow regime of the GRB represents an intermittent/ephemeral system, particularly over the last decade (2000-2012) (Taie Semiromi & Koch, 2019), when baseflow contributes to streamflow only in a sporadic and irregular way. In other words, GRB's river system characterized originally by a perennial system has been converting into an intermittent system due to a depletion of the groundwater aquifer with a subsequent drop of the groundwater table, that is, decreasing head differences with the adjacent gaining streams. Furthermore, and more importantly, as surface runoff (but not lateral flow), as the main constituent of the water yield (see equation 3) was shown earlier (see Figure 8) to be not affected by any of the two pumping variants, it is of no surprise that the future water yield will not be impacted neither. Nonetheless, in conclusion of this section, as the surface runoff will generally diminish in the wake of the climate change projected for the various RCP emission scenarios (see Figure 8), less water will be yielded in the future than in the historic period as well. And, counterintuitively, this future adverse situation cannot be remedied neither





Figure 19. Similar to Figure 17, but for the monthly total water yield of the GRB.

by shutting off the groundwater pumping wells, as this was shown to have no effect on the future projected (decreasing) surface runoff (see Figure 8) which, in turn, will lead to the diminished future water yield seen in Figures 18 and 19.

5. Summary and Conclusions

Most water resources studies carried out to date have been directed towards predicting the potential impacts of future climate change on surface water hydrology, while less attention has been paid to such impacts on groundwater hydrology. And even less as of yet are researches geared towards a better understanding of the net effects of the climate change on the fate of gaining and losing streams in a watershed/ aquifer, as this requires a coupled, more complex approach to the intertwined surface- and subsurface hydrological compartments. In the light of this scarcity of appropriate studies regarding this topic, the present study aims, therefore, to shed light on the different processes acting in both the surface- and subsurface water compartment, including, in particular the complex hydrological interactions of the surface and groundwater domain, as they are mainly reflected by the processes taking place in the losing and gaining streams. This task is achieved through the application of the fully coupled surface-subsurface hydrological model SWAT-MODFLOW-NWT to the Gharehsoo River Basin (GRB), with the enclosed Ardabil aquifer, located in northwestern Iran. This coupled hydrological model was forced with the downscaled climate predictors (scenarios) under three representative concentration pathways (RCPs), namely, RCP 2.6, 4.5, and 8.5 for up to year 2100, dividing the whole simulation period in a reference historic period (1978-2012) and three future time slices 2013-2040, 2041-2070, and 2071-2100. Furthermore, to distinguish the influence of the prevalent groundwater overdraft in the GRB on the gaining and losing streams from that of climate change, we also run the climate-forced coupled hydrological model under two future pumping variants, that is, "nonpumping" (assuming a shut-off of all groundwater



extraction at the end of the historic period) and "pumping" (assuming an ongoing of the average groundwater extraction of the recent past also in the future).

First of all, the results of temperature downscaling indicate that a significant rise in monthly min temperature, particularly during winter (e.g., January and February) under all scenarios, wherefore this increase will become more tangible for RCP 8.5 and the distant future (2071–2100) time slice. In contrast, for the future months of September a minor decrease in min temperature is observed for all stations, except Sareyn.

Similarly, the prediction of monthly maximum temperature demonstrates that a considerable future increase for all months, except April at all weather stations. Similar to the downscaled min temperature, the max temperature will also soar dramatically when going from RCP 2.6 to 8.5 and from the near future, 2006–2040, to the distant future, 2071–2100. Overall, the lowest increases of the annual average max temperature of 2.87 °C under RCP 2.6 and time slice 2006–2040 to up to 8.54 °C under RCP 8.5 and time slice 2071–2100 are projected for the Sareyn station, whereas the greatest increases of 3.65 °C under RCP 2.6 and time slice 2006–2040 up to 9.7 °C under RCP 8.5 and time slice 2071–2100 will occur at Nir station.

The projections of the precipitation using a recently developed hybrid model DWT-ANN-QM of the authors (Taie Semiromi & Koch, 2017) point out that the daily precipitation under RCP 2.6, 4.5, and 8.5 will, relative to the reference period (1978–2005), slightly decrease for the 2006–2040 time slice, slightly increase for 2041–2070, and decrease again for 2071–2100, resulting overall in an average decrease of about 12% across the GRB by year 2100. Interestingly, this percentage reduction of the precipitation will be more under RCP 4.5 than under RCP 8.5, owing most likely to the fact that the higher temperatures anticipated for the later than the former will bring about higher evapotranspiration, that is, more water vapor into the atmosphere and, thus, will be higher propensity for precipitation under RCP 8.5. On the other hand, for all RCPs, the monthly precipitation will increase in future months of the wet period December–February, meanwhile for all the other months a decrease, except under RCP 8.5 and somehow 4.5, is expected. Using these downscaled climate predictors in the SWAT-MODFLOW-NWT coupled hydrological model, the various water budgets components of the surface-subsurface compartment of the hydrological cycle (surface runoff, diffuse surface-to-groundwater recharge, baseflow, etc.) were projected into the future.

For example, regarding the diffuse groundwater recharge, due to the projected rise in the future precipitation during the winter, an increase of this quantity, relative to the historic period, will also occur across all future periods considered, 2013–2040, 2040–2070, and 2071–2099 and this holds for all three RCPs (2.6, 4.5 and 8.5). In the other dry season months (e.g., July to October), the diffuse groundwater recharge will be the same as that of the historic period (except for distant future period 2071–2099), because the soil water content will be very low during that season, so that even in case of a storm event, most of the infiltrated water will be trapped due to a very high matric potential of the soil; hence, inhibiting diffuse groundwater recharge. However, the latter is expected to decrease in some future months, such as March under RCP 8.5 and May under all three RCPs. In addition to the change of the magnitude of the recharge, its peak will also be shifted from March to April owing to the anticipated future winter warming. The SWAT-MODFLOW-NWT coupled surface-groundwater model analysis of the GRB of Taie Semiromi and Koch (2019) indicates that the major source of the groundwater recharge for the Ardabil aquifer in the recent past is the concentrated recharge generated from the losing streams, corroborating previous qualitative estimations of Ardabil Regional Water Authority (2013). In contrast, the rate of the present-day meteoric groundwater recharge (diffuse recharge) is with only ~12 mm/yr (see Figure 7 and Taie Semiromi and Koch (2019), for details) rather low, due to (1) the prevailing semiarid climate with only ~270/yr mm precipitation which leads to a very high soil matric potential in most of the seasons, so that a major fraction of the total infiltration (gross recharge) is retained in the vadose zone, in turn, leaves less water for the deeper net groundwater recharge; (2) the build-up, impervious surface area of Ardabil city, as the capital of the same province, covers a noticeable fraction of the aquifer's total area (around 70 km²), so that diffuse surface-to-groundwater recharge is almost negligible. To quantify the range of uncertainty of the estimated groundwater recharge rate for this aquifer, Healy (2010) suggested that this hydrological variable should be evaluated using different approaches. Although, on average, 89% of the total water consumption in the GRB is supplied from groundwater resources (K. Kord & Asghari Moghaddam, 2014), knowledge of the Ardabil aquifer's hydrogeological characteristics, including recharge rate, suffers from a lack of detailed exploration. Thus, our study here can be considered a supporting research task in that regard that can enhance the understanding of the various hydrological processes controlling the interaction of the surface- and subsurface water compartments in the GRB.

The SWAT-MODFLOW-NWT-predictions of the future climate change impacts on the surface-groundwater interactions demonstrates that the baseflow generated from the groundwater discharge to the gaining streams will be more influenced than the concentrated groundwater recharge from the losing streams. However, more importantly, the future fate of the baseflow depends tremendously on the pumping variant used in the model. Thus, whereas for the "nonpumping" variant (shutting off all wells after year 2013) the baseflow will significantly increase in the future - though tapered noticeably by adverse effects of climate change (less precipitation) in the distant future time period 2071-2099-, the opposite is true for the "pumping" variant (which assumes that the groundwater utilization for the whole projection period (2013–2099) will remain the same as that of the last year of the historic period, 2012), that is, a notable ongoing decrease of the baseflow is obtained under all RCPs, again more strongly in the distant future period. These projected changes of the baseflow affect also the future baseflow index (BI), though, because of its definition (ratio of baseflow to total streamflow), somewhat differently as baseflow itself. Thus, BI rises spectacularly in the future, irrespective of the pumping variant employed. This is due to the dominant effect of the surface runoff (in the denominator of the BI-equation) which will strongly decrease in the future due to climate change (decreasing precipitation), and this is, of course, not affected by the pumping variant. In contrast to the high sensitivity of the future baseflow to the two pumping variants, this does not hold for the concentrated groundwater recharge produced by the losing (influent) streams. First of all, the total amount of concentrated recharge in the GRB is up to nearly two orders of magnitude smaller than the previously discussed concentrated discharge (baseflow). This is due to the fact that since most of the area of the modeled Ardabil aquifer is rather flat plain, where the groundwater table is near the land surface, positive head difference towards the river gage heights are produced, which leads to the sizable generation of baseflow. Furthermore, inasmuch as the future surface runoff was shown to be considerably abated under the different climate forcing scenarios, less streamflow will also be available within the river network for the concentrated groundwater recharge from possibly losing streams. It is interesting to note that, in contrast to the strong future projected augmentation of the baseflow under the "nonpumping" variant, the future concentrated groundwater recharge from the losing streams will be slightly reduced, because the future groundwater table is expected to fall less under this "nonpumping" variant and due to an additional small future increase of the diffuse surface-to-groundwater recharge (see Figure 7), which could cause a reversal of the head gradient between the river and the unconfined upper aquifer layer, so that, eventually, a losing stream (influent conditions) may switch to a gaining stream (effluent condition).

In addition to the temporal assessment of the climate change and pumping impacts on the losing and gaining streams, we investigated also their spatial variability. On this subject, the results illustrate that the maximum daily baseflow originating from all gaining streams will rise under all of the three climatic emission scenarios (RCPs) as well as for the two pumping variants, for instance, from 622.71 m³/day, for the historic period to 709.29 m³/day (in the upstream, hilly sections of the GBR) for time slice 2041–2070 (during which an increase of the precipitation is projected) under RCP 2.6 and "nonpumping". In spite of these sporadic and local increases, the future total daily baseflow will slightly decrease for the "pumping" variant, and this holds for all three RCPs. On the contrary, for the "nonpumping" variant the daily baseflow of $63415 \text{ m}^3/\text{day}$ obtained for the historic period will soar to values higher than 70000 m³/day, depending on the time slice, but irrespective of the RCP emission scenario.

Regarding the spatial variability of the concentrated groundwater recharge from the losing streams, similarly to its future temporal reactions to the climate change scenarios and the pumping variants mentioned earlier, only a low sensitivity of this parameter can be seen in the corresponding maps, unlike for the baseflow. Nevertheless, the future total daily volume of the concentrated groundwater recharge for the whole aquifer will considerably change under the various combinations of emission scenarios and pumping variants. Despite the fact that the number of losing streams, 1,659, is much higher than that of the gaining streams, 922, the concentrated groundwater recharge from the former is up to two orders of magnitude smaller than the groundwater discharge to the latter. The maximum of the "little" groundwater recharge from the losing streams of only 0.4 m³/day will occur over the western part of the Ardabil aquifer during the middle

future (2041–2070), under RCP 2.6 and for the "pumping" variant. And dissimilar to the baseflow, the future total daily concentrated groundwater recharge volume over the aquifer is anticipated to rise under the "pumping" variant, irrespective of the RCP emission scenarios. For the "nonpumping" variant, on the contrary, the future total concentrated groundwater recharge from the losing streams will drop considerably, due to a lesser falling of the groundwater heads and the additional impact of a slightly increased future diffuse surface-to-groundwater recharge (see Figure 7), that forces some of the losing streams to converted to gaining streams.

Regarding the groundwater storage projected under the climatic emission scenarios and the two pumping variants, the results point out that the present-day unsustainable groundwater abstraction (continued into the future in the "pumping" variant) is the real agent driving for an ongoing drawdown, that is, groundwater storage decrease, in the future, even if the 2622 farming pumping wells that extract around 150 MCM water per year are completely shut off after the end of the historic period (2012). And not only that, the initial groundwater storage that existed in the initial years (initial condition of the simulation) of the historic period (e.g., 1980) cannot even be recovered, as the small absolute amount of the future diffuse groundwater recharge, albeit slightly increasing from historic 10 mm/yr to future 11 mm/yr, for RCP 2.6 and 4.5 (though not for RCP 8.5, because of less precipitation and higher evapotranspiration) (see Figure 7), is not able to replenish the depleted aquifer. In conclusion, the delicate balance between the total groundwater recharge and discharge for the Ardabil aquifer in the predevelopment has been irrevocably lost over the last two decades and will continue to do so even more in the future, accentuated by the adverse impacts of climate change (reduced precipitation in the near future, together with a small degree of diffuse surface-to-groundwater recharge in the GRB).

Eventually, the effects of the different climate emission scenarios and pumping variants on the total water yield of the GRB are determined. The results show that for both pumping variants, this quantity will drop for all future months, except for April in time slice 2041–2070, under the (benevolent) RCP 2.6, the time period for which an increase in the precipitation is projected. Surprisingly, "nonpumping" does not noticeably improve this adverse future water yield situation. This is, because the major fraction of the water yield is contributed by the surface runoff in the GRB and, for this reason, the same findings for the latter are obtained for the water yield as well, that is, a shift of its apex from historic-time March to April. Overall, less water will be available in the GRB in the future than in the historic period, wherefore, expectedly, this adverse effect will be the strongest for the most extreme RCP 8.5 climate emission scenario.

Although the results of the present study indicate that by combining a complex semidistributed surface water model (SWAT) with a 3D groundwater model (MODFLOW-NWT) significant enhancements in the understanding of the interaction of surface-subsurface hydrological processes can be achieved, there are several remaining, general and particular issues yet, that should be addressed in further studies. First of all, one concern is the general uncertainty attributed to the parameterization, parameter estimation, and to the observations required for setting up, driving and evaluating a hydrological model need (Wagener & Gupta, 2005; Y. Liu & Gupta, 2007). This is particularly true for integrated hydrological models such as SWAT-MODFLOW-NWT used here which requires a large number of input parameters related to the land surface, vadose zone, and groundwater domain hydrological processes, many of which could only be prescribed here on an ad-hoc, experience-based manner (Bailey et al., 2016; Taie Semiromi & Koch, 2019). Thus, in spite of our awareness of the ensuing uncertainties in the simulated surface-groundwater interactions, and so impacting the overall water balance in the study basin (GRB) region, such a general uncertainty analysis has not been performed in the present study. Suffice to say here that the hydraulic conductance of the stream-aquifer interface (river banks and bottom beds) appears to have a pivotal role in that regard.

Secondly, since future climate change scenarios are also faced with large uncertainties, the effects of latter on the surface-groundwater interactions should also be investigated. Thirdly, future land-use and land-cover change (LULCC), not considered in the present study, although certainly being subject to change by 2100, will have additional implications for the water resources budget of the GRB. The most recent SWAT-model version allows to implement updated LULCC during the ongoing simulation (Moriasi et al., 2019) and future investigations should take this into consideration. With these reservations, the water resources results obtained for the GBR in the present study should be seen more as trends rather than exact predictions. Notwithstanding these caveats, as the climate change impacts on the water components were



projected under both current (historic) groundwater utilization ("pumping") and "nonpumping" variants, the here proposed fully coupled surface-subsurface hydrological model, that is, SWAT-MODFLOW-NWT, can make it possible to estimate the maximally sustainable groundwater utilization amount. Consequently, the adverse impacts of future climate change on the water resources can be alleviated. The influence of most likely occurring future increments of the groundwater demand was not considered in the current study, because Ardabil Water Authority has prohibited further groundwater utilization from the Ardabil aquifer for the purpose of recovering the ongoing sharp groundwater head drawdown. Ergo, model adjustments to that regard are recommended in future studies, since there is no doubt that the groundwater demand will rise in the future, not only due to a shortage of surface water (triggered by a decrease in precipitation and increase in temperature), but also due to population growth as well as expanded irrigated agriculture.

Although the findings of the present study may not be comparable to other hydrogeomorphic regions in the world because temperature and, particularly, precipitation vary spatially to a large extent over the globe, our proposed hybrid modeling scheme can successfully be applied to any other study region for future water resources management and planning in the wake of different anthropocentric and hydroclimate stressors, provided that GCMs and/or RCM-climate predictors under different GHG emission scenarios are available which are, however, new sources of uncertainty.

Conflict of Interest

The authors declare no conflict of interest.

References

- Alvarez-Villa, O. D., Cassiraga, E., & Sahuquillo, A. (2014). Towards a more efficient simulation of surface-groundwater interaction in conjunctive use systems: Selective compression and modal masking in the eigenvalue method. *Mathematics of Planet Earth*, 439–442. https://doi.org/10.1007/978-3-642-32408-6_97
- Anderson, M. P., Woessner, W. W., & Hunt, R. J. (2015). Applied groundwater modeling: Simulation of flow and advective transport, (2nd ed.). New York, USA: Elsevier.
- Ardabil Regional Water Authority. (2013). Investigation of groundwater balance in Ardabil plain. Retrieved from Ardabil, Iran:
- Arnold, J. G., Allen, P. M., & Bernhardt, G. (1993). A comprehensive surface-groundwater flow model. *Journal of Hydrology*, 142(1-4), 47–69. https://doi.org/10.1016/0022-1694(93)90004-S
- Arnold, J. G., Srinivasan, R., Muttiah, R. S., & William, J. R. (1998). Large area hydrologic modeler and assessment part I: Model development. [Large area hydrologic modeling and assessment part 1: Model development]. Journal of the American Water Resources Association, 34(1), 73–89.
- Ashwell, N. E. Q., Peterson, J. M., & Hendricks, N. P. (2017). Optimal groundwater management under climate change and technical progress. *Resource and Energy Economics*, 51, 67–83. https://doi.org/10.1016/j.reseneeco.2017.10.005
- Baalousha, H. M. (2012). Modelling surface-groundwater interaction in the Ruataniwha basin, Hawke's Bay, New Zealand. Environmental Earth Sciences, 66(1), 285–294. https://doi.org/10.1007/s12665-011-1238-y
- Bailey, R. T., Wible, T. C., Arabi, M., Records, R. M., & Ditty, J. (2016). Assessing regional-scale spatio-temporal patterns of groundwater-surface water interactions using a coupled SWAT-MODFLOW model. *Hydrological Processes*, 30(23), 4420–4433. https:// doi.org/10.1002/hyp.10933
- Bejranonda, W., Koontatakulvong, S., & Koch, M. (2007, September 17–21, 2007), Surface and groundwater dynamic interactions in the Upper Great Chao Phraya Plain of Thailand: Semi-coupling of SWAT and MODFLOW. Paper presented at the IAH 2007, Groundwater and Ecosystems, Lisbon 2007, Lisbon, Portugal.
- Bovolo, C. I., Parkin, G., & Sophocleous, M. (2009). Groundwater resources, climate and vulnerability. *Environmental Research Letters*, 4(3). https://doi.org/10.1088/1748-9326/4/3/035001
- Brunner, P., & Simmons, C. T. (2012). HydroGeoSphere: A fully integrated, physically based hydrological model. *Groundwater*, 50(2), 170–176. https://doi.org/10.1111/j.1745-6584.2011.00882.x
- Camporese, M., Paniconi, C., Putti, M., & Orlandini, S. (2010). Surface-subsurface flow modeling with path-based runoff routing, boundary condition-based coupling, and assimilation of multisource observation data. Water Resources Research, 46, W02512. https://doi.org/ 10.1029/2008WR007536
- Chen, J. W., Kuo, H. T., & Lin, H. L. (2016). Estimation the hot spring recharge using groundwater numerical model in Jiao-Shi area, Taiwan. Proceedings of the Ieee International Conference on Advanced Materials for Science and Engineering (Ieee-Icamse 2016), 434-437. Retrieved from <go to ISI>://WOS:000402015500123
- Chen, Z. H., Grasby, S. E., & Osadetz, K. G. (2002). Predicting average annual groundwater levels from climatic variables: An empirical model. *Journal of Hydrology*, 260(1-4), 102–117. https://doi.org/10.1016/S0022-1694(01)00606-0
- Cousquer, Y., Pryet, A., Flipo, N., Delbart, C., & Dupuy, A. (2017). Estimating river conductance from prior information to improve surface-subsurface model calibration. *Groundwater*, 55(3), 408–418. https://doi.org/10.1111/gwat.12492
- Crosbie, R. S., McCallum, J. L., Walker, G. R., & Chiew, F. H. S. (2010). Modelling climate-change impacts on groundwater recharge in the Murray-Darling Basin, Australia. *Hydrogeology Journal*, 18(7), 1639–1656. https://doi.org/10.1007/s10040-010-0625-x
- Crosbie, R. S., Pickett, T., Mpelasoka, F. S., Hodgson, G., Charles, S. P., & Barron, O. V. (2013). An assessment of the climate change impacts on groundwater recharge at a continental scale using a probabilistic approach with an ensemble of GCMs. *Climatic Change*, 117(1–2), 41–53. https://doi.org/10.1007/s10584-012-0558-6

Acknowledgments

The authors would like to thank Deutscher Akademischer Austauschdienst (DAAD) for awarding a scholarship to the first author to support this project under the funding program/-ID: Sustainable Water Management: Study Scholarships and Research Grants 14 (56322373). Datasets used and created for this research are available in Taie Semiromi and Koch (2019) and Taie Semiromi and Koch (2017).

- Daggupati, P., Pai, N., Ale, S., Douglas-Mankin, K. R., Zeckoski, R. W., Jeong, J., et al. (2015). A recommended calibration and validation strategy for hydrologic and water quality models. *Transactions of the Asabe*, 58(6), 1705–1719.
- Dwire, K. A., Mellmann-Brown, S., & Gurrieri, J. T. (2017). Potential effects of climate change on riparian areas, wetlands, and groundwater-dependent ecosystems in the Blue Mountains, Oregon, USA. *Climate Services*, 10, 44–52. https://doi.org/10.1016/j. cliser.2017.10.002
- Ebel, B. A., Mirus, B. B., Heppner, C. S., VanderKwaak, J. E., & Loague, K. (2009). First-order exchange coefficient coupling for simulating surface water-groundwater interactions: Parameter sensitivity and consistency with a physics-based approach. *Hydrological Processes*, 23(13), 1949–1959. https://doi.org/10.1002/hyp.7279
- Erturk, A., Ekdal, A., Gurel, M., Karakaya, N., Guzel, C., & Gonenc, E. (2014). Evaluating the impact of climate change on groundwater resources in a small Mediterranean watershed. Science of the Total Environment, 499, 437–447. https://doi.org/10.1016/j. scitotenv.2014.07.001
- Furman, A. (2008). Modeling coupled surface-subsurface flow processes: A review. Vadose Zone Journal, 7(2), 741–756. https://doi.org/ 10.2136/vzj2007.0065
- Gemitzi, A., Ajami, H., & Richnow, H. H. (2017). Developing empirical monthly groundwater recharge equations based on modeling and remote sensing data—Modeling future groundwater recharge to predict potential climate change impacts. *Journal of Hydrology*, 546, 1–13. https://doi.org/10.1016/j.jhydrol.2017.01.005
- Ghazavi, R., & Ebrahimi, H. (2019). Predicting the impacts of climate change on groundwater recharge in an arid environment using modeling approach. International Journal of Climate Change Strategies and Management, 11(1), 88–99. https://doi.org/10.1108/ijccsm-04-2017-0085
- Goderniaux, P., Brouyere, S., Blenkinsop, S., Burton, A., Fowler, H. J., Orban, P., & Dassargues, A. (2011). Modeling climate change impacts on groundwater resources using transient stochastic climatic scenarios. Water Resources Research, 47, W12516. https://doi.org/10.1029/ 2010wr010082
- Goderniaux, P., Brouyère, S., Fowler, H. J., Blenkinsop, S., Therrien, R., Orban, P., & Dassargues, A. (2009). Large scale surface–subsurface hydrological model to assess climate change impacts on groundwater reserves. *Journal of Hydrology*, 373(1), 122–138. https://doi.org/ 10.1016/j.jhydrol.2009.04.017
- Goderniaux, P., Brouyere, S., Wildemeersch, S., Therrien, R., & Dassargues, A. (2015). Uncertainty of climate change impact on groundwater reserves—Application to a chalk aquifer. *Journal of Hydrology*, 528, 108–121. https://doi.org/10.1016/j. jhydrol.2015.06.018
- Goodarzi, M., Abedi-Koupai, J., & Heidarpour, M. (2019). Investigating impacts of climate change on irrigation water demands and its resulting consequences on groundwater using CMIP5 models. *Groundwater*, *57*(2), 259–268. https://doi.org/10.1111/gwat.12771
- Goodarzi, M., Abedi-Koupai, J., Heidarpour, M., & Safavi, H. R. (2016). Evaluation of the effects of climate change on groundwater recharge using a hybrid method. *Water Resources Management*, 30(1), 133–148. https://doi.org/10.1007/s11269-015-1150-4
- Green, T. R., Taniguchi, M., & Kooi, H. (2007). Potential impacts of climate change and human activity on subsurface water resources. Vadose Zone Journal, 6(3), 531–532. https://doi.org/10.2136/vzj2007.0098
- Green, T. R., Taniguchi, M., Kooi, H., Gurdak, J. J., Allen, D. M., Hiscock, K. M., et al. (2011). Beneath the surface of global change: Impacts of climate change on groundwater. *Journal of Hydrology*, 405(3–4), 532–560. https://doi.org/10.1016/j.jhydrol.2011.05.002
- Havril, T., Tóth, Á., Molson, J. W., Galsa, A., & Mádl-Szőnyi, J. (2017). Impacts of predicted climate change on groundwater flow systems: Can wetlands disappear due to recharge reduction? *Journal of Hydrology*, 563, 1169–1180. https://doi.org/10.1016/j. jhydrol.2017.09.020
- Hayashi, M., & Farrow, C. R. (2014). Watershed-scale response of groundwater recharge to inter-annual and inter-decadal variability in precipitation (Alberta, Canada). *Hydrogeology Journal*, 22(8), 1825–1839. https://doi.org/10.1007/s10040-014-1176-3
- Healy, R. W. (2010). Estimating groundwater recharge. Cambridge: Cambridge University Press.
- Holman, I. P. (2006). Climate change impacts on groundwater recharge-uncertainty, shortcomings, and the way forward? Hydrogeology Journal, 14(5), 637–647. https://doi.org/10.1007/s10040-005-0467-0
- Hu, Y. R., Maskey, S., & Uhlenbrook, S. (2013). Downscaling daily precipitation over the Yellow River source region in China: A comparison of three statistical downscaling methods. *Theoretical and Applied Climatology*, 112(3–4), 447–460. https://doi.org/10.1007/ s00704-012-0745-4
- Huang, A. N., Zhou, Y., Zhang, Y. C., Huang, D. Q., Zhao, Y., & Wu, H. M. (2014). Changes of the annual precipitation over Central Asia in the twenty-first century projected by multimodels of CMIP5. *Journal of Climate*, 27(17), 6627–6646. https://doi.org/10.1175/Jcli-D-14-00070.1
- Huber, U. M., Bugmann, H. K. M., & Reasoner, M. A. (2005). Global change and mountain regions: An overview of current knowledge. Dordrecht: Springer.
- IPCC (2013). IPCC fifth assessment report. Weather, 68(12), 310-310.
- Irvine, D. J., Cranswick, R. H., Simmons, C. T., Shanafield, M. A., & Lautz, L. K. (2015). The effect of streambed heterogeneity on groundwater-surface water exchange fluxes inferred from temperature time series. Water Resources Research, 51, 198–212. https://doi. org/10.1002/2014wr015769
- Javan, K., Lialestani, M. R. F. H., Ashouri, H., & Moosavian, N. (2015). Assessment of the impacts of nonstationarity on watershed runoff using artificial neural networks: A case study in Ardebil, Iran. *Modeling Earth Systems and Environment*, 22(1). https://doi.org/10.1007/s40808-015-0030-5
- Jellalia, D., Lachaal, F., Andoulsi, M., Zouaghi, T., Hamdi, M., & Bedir, M. (2015). Hydro-geophysical and geochemical investigation of shallow and deep Neogene aquifer systems in Hajeb Layoun-Jilma-Ouled Asker area, Central Tunisia. *Journal of African Earth Sciences*, *110*, 227–244. https://doi.org/10.1016/j.jafrearsci.2015.06.016
- Jyrkama, M. I., & Sykes, J. F. (2007). The impact of climate change on spatially varying groundwater recharge in the grand river watershed (Ontario). *Journal of Hydrology*, 338(3–4), 237–250. https://doi.org/10.1016/j.jhydrol.2007.02.036
- Kahsay, K. D., Pingale, S. M., & Hatiye, S. D. (2018). Impact of climate change on groundwater recharge and base flow in the sub-catchment of Tekeze basin, Ethiopia. *Groundwater for Sustainable Development*, *6*, 121–133. https://doi.org/10.1016/j.gsd.2017.12.002
- Kalbus, E., Reinstorf, F., & Schirmer, M. (2006). Measuring methods for groundwater-surface water interactions: A review. *Hydrology and Earth System Sciences*, 10(6), 873–887.
- Karian, Z. A., & Dudewicz, E. J. (2000). Fitting statistical distributions: The generalized lambda distribution and generalized bootstrap methods. Boca Raton: CRC Press.
- Kim, K. B., Kwon, H. H., & Han, D. W. (2018). Exploration of warm-up period in conceptual hydrological modelling. Journal of Hydrology, 556, 194–210. https://doi.org/10.1016/j.jhydrol.2017.11.015

- Kirby, J. M., Mainuddin, M., Mpelasoka, F., Ahmad, M. D., Palash, W., Quadir, M. E., et al. (2016). The impact of climate change on regional water balances in Bangladesh. *Climatic Change*, 135(3-4), 481–491. https://doi.org/10.1007/s10584-016-1597-1
- Klove, B., Ala-Aho, P., Bertrand, G., Gurdak, J. J., Kupfersberger, H., Kværner, J., et al. (2014). Climate change impacts on groundwater and dependent ecosystems. *Journal of Hydrology*, 518, 250–266. https://doi.org/10.1016/j.jhydrol.2013.06.037
- Kolditz, O., Bauer, S., Bilke, L., Böttcher, N., Delfs, J. O., Fischer, T., et al. (2012). OpenGeoSys: An open-source initiative for numerical simulation of thermo-hydro-mechanical/chemical (THM/C) processes in porous media. *Environmental Earth Sciences*, 67(2), 589–599. https://doi.org/10.1007/s12665-012-1546-x
- Kord, K., & Asghari Moghaddam, A. (2014). Spatial analysis of Ardabil plain aquifer potable groundwater using fuzzy logic. Journal of King Saud University - Science, 129(26), 129–140.
- Kord, M., Asghari Moghaddam, A., & Nakhaeei, M. (2013). Investigation of hydrogeological characteristics of Ardabil plain aquifer, northwest of Iran. ISESCO JOURNAL of Science and Technology, 9(15), 63–69.
- Liu, W., Bailey, R. T., Andersen, H. E., Jeppesen, E., Park, S., Thodsen, H., et al. (2020). Assessing the impacts of groundwater abstractions on flow regime and stream biota: Combining SWAT-MODFLOW with flow-biota empirical models. *Science of the Total Environment*, 706, 135,702. https://doi.org/10.1016/j.scitotenv.2019.135702
- Liu, Y., & Gupta, H. V. (2007). Uncertainty in hydrologic modeling: Toward an integrated data assimilation framework. Water Resources Research, 43, W07401. https://doi.org/10.1029/2006wr005756
- Mani, A., Tsai, F. T. C., Kao, S. C., Naz, B. S., Ashfaq, M., & Rastogi, D. (2016). Conjunctive management of surface and groundwater resources under projected future climate change scenarios. *Journal of Hydrology*, 540, 397–411. https://doi.org/10.1016/j. ihydrol.2016.06.021
- Maxwell, R. M., Chow, F. K., & Kollet, S. J. (2007). The groundwater–land-surface–atmosphere connection: Soil moisture effects on the atmospheric boundary layer in fully-coupled simulations. Advances in Water Resources, 30(12), 2447–2466. https://doi.org/10.1016/j. advwatres.2007.05.018
- McDonald, M. G., Harbaugh, A. W., & geological survey (U.S.). (1988). A modular three-dimensional finite-difference ground-water flow model. Reston, Va.?: U.S. Geological Survey.
- Melo, D. C. D., & Wendland, E. (2017). Shallow aquifer response to climate change scenarios in a small catchment in the Guarani aquifer outcrop zone. Anais Da Academia Brasileira De Ciencias, 89(1), 391–406. https://doi.org/10.1590/0001-3765201720160264
- Meyboom, P. (1961). Estimating ground-water recharge from stream hydrographs. *Journal of Geophysical Research*, 66(4), 1203–1214. https://doi.org/10.1029/JZ066i004p01203
- Min, L., Shen, Y., & Pei, H. (2015). Estimating groundwater recharge using deep vadose zone data under typical irrigated cropland in the piedmont region of the North China Plain. *Journal of Hydrology*, 527, 305–315. https://doi.org/10.1016/j.jhydrol.2015.04.064
- Modarres, R., Ghadami, M., Naderi, S., & Naderi, M. (2018). Future extreme rainfall change projections in the north of Iran. *Meteorological Applications*, 25(1), 40–48. https://doi.org/10.1002/met.1667
- Modica, E., & Buxton, H. T. (1998). Evaluating the source and residence times of groundwater seepage to streams, New Jersey Coastal Plain. *Water Resources Research*, 34(11), 2797–2810. https://doi.org/10.1029/98WR02472
- Moeck, C., Brunner, P., & Hunkeler, D. (2016). The influence of model structure on groundwater recharge rates in climate-change impact studies. *Hydrogeology Journal*, 24(5), 1171–1184. https://doi.org/10.1007/s10040-016-1367-1
- Mohamed, A. S., Leduc, C., Marlin, C., Wague, O., & Cheikh, M. A. S. (2017). Impacts of climate change and anthropization on groundwater resources in the Nouakchott urban area (coastal Mauritania). *Comptes Rendus Geoscience*, 349(6–7), 280–289. https://doi.org/ 10.1016/j.crte.2017.09.011
- Moriasi, D. N., Pai, N., Steiner, J. L., Gowda, P. H., Winchell, M., Rathjens, H., et al. (2019). SWAT-LUT: A desktop graphical user interface for updating land use in SWAT. Journal of the American Water Resources Association, 55(5), 1102–1115. https://doi.org/10.1111/1752-1688.12789
- Nassery, H., & Salami, H. (2016). Identifying vulnerable areas of aquifer under future climate change (case study: Hamadan aquifer, West Iran). Arabian Journal of Geosciences, 9(8), 1–16. https://doi.org/10.1007/s12517-016-2526-3
- Neitsch, S. L., Arnold, J. G., Kiniry, J. R., & Williams, J. R. (2011). Soil and water assessment tool theoretical documentation version 2009. Retrieved from Texas Water Resources Institute, College Station, Texas: https://swat.tamu.edu/media/99192/swat2009-theory. pdf
- Nistor, M. M., & Mindrescu, M. (2019). Climate change effect on groundwater resources in Emilia-Romagna region: An improved
- assessment through NISTOR-CEGW method. *Quaternary International*, 504, 214–228. https://doi.org/10.1016/j.quaint.2017.11.018 Niswonger, R. G., Panday, S., & Motomu, I. (2011). MODFLOW-NWT, a Newton formulation for MODFLOW-2005: U.S. Geological Survey Techniques and Methods 6-A37. Retrieved from
- Nkhonjera, G. K., & Dinka, M. O. (2017). Significance of direct and indirect impacts of climate change on groundwater resources in the Olifants River basin: A review. *Global and Planetary Change*, 158, 72–82. https://doi.org/10.1016/j.gloplacha.2017.09.011
- Nourani, V., Alami, M. T., & Vousoughi, F. D. (2015). Wavelet-entropy data pre-processing approach for ANN-based groundwater level modeling. *Journal of Hydrology*, 524, 255–269. https://doi.org/10.1016/j.jhydrol.2015.02.048
- Park, S., & Bailey, R. T. (2017). SWAT-MODFLOW tutorial: Documentation for preparing model simulations. Retrieved from Fort Collins, Colorado, USA:
- Riasati, A., McFarlane, D., Varma, S., Dawes, W., Emelyanova, I., Hodgson, G., & Charles, S. (2012). Potential climate change impacts on groundwater resources of south-western Australia. Journal of Hydrology, 475, 456–472. https://doi.org/10.1016/j.jhydrol.2012.04.043
- Rosenberg, N. J., Epstein, D. J., Wang, D., Vail, L., Srinivasan, R., & Arnold, J. G. (1999). Possible impacts of global warming on the hydrology of the Ogallala aquifer region. *Climatic Change*, *42*(4), 677–692. https://doi.org/10.1023/A:1005424003553
- Rushton, K. (2007). Representation in regional models of saturated river-aquifer interaction for gaining/losing rivers. *Journal of Hydrology*, 334(1–2), 262–281. https://doi.org/10.1016/j.jhydrol.2006.10.008
- Saha, G. C., Li, J., Thring, R. W., Hirshfield, F., & Paul, S. S. (2017). Temporal dynamics of groundwater-surface water interaction under the effects of climate change: A case study in the Kiskatinaw River Watershed, Canada. *Journal of Hydrology*, 551, 440–452. https://doi.org/10.1016/j.jhydrol.2017.06.008
- Sanderson, M. R., & Curtis, A. L. (2016). Culture, climate change and farm-level groundwater management: An Australian case study. Journal of Hydrology, 536, 284–292. https://doi.org/10.1016/j.jhydrol.2016.02.032
- Scibek, J., & Allen, D. M. (2006). Modeled impacts of predicted climate change on recharge and groundwater levels. Water Resources Research, 42, W11405. https://doi.org/10.1029/2005wr004742

- Scibek, J., Allen, D. M., Cannon, A. J., & Whitfield, P. H. (2007). Groundwater–surface water interaction under scenarios of climate change using a high-resolution transient groundwater model. *Journal of Hydrology*, 333(2), 165–181. https://doi.org/10.1016/j. jhydrol.2006.08.005
- Shi, X. M., & Durran, D. R. (2014). The response of orographic precipitation over idealized midlatitude mountains due to global increases in CO₂. Journal of Climate, 27(11), 3938–3956. https://doi.org/10.1175/Jcli-D-13-00460.1
- Shrestha, S., Bach, T. V., & Pandey, V. P. (2016). Climate change impacts on groundwater resources in Mekong Delta under representative concentration pathways (RCPs) scenarios. *Environmental Science & Policy*, 61, 1–13. https://doi.org/10.1016/j.envsci.2016.03.010
- Smerdon, B. D. (2017). A synopsis of climate change effects on groundwater recharge. Journal of Hydrology, 555, 125–128. https://doi.org/ 10.1016/i.ihvdrol.2017.09.047
- Statistical Center of Iran (2011). Implementation of the 2011 Iranian Population and Housing Census. Iran: Statistical Center of Iran Retrieved from https://www.amar.org.ir/english
- Stoll, S., Franssen, H. J. H., Barthel, R., & Kinzelbach, W. (2011). What can we learn from long-term groundwater data to improve climate change impact studies? *Hydrology and Earth System Sciences*, 15(12), 3861–3875. https://doi.org/10.5194/hess-15-3861-2011
- Taie Semiromi, M., & Koch, M. (2017). Downscaling of daily precipitation using a hybrid model of artificial neural network, wavelet, and quantile mapping in Gharehsoo River Basin, Iran. Paper Presented at the American Geophysical Union (AGU) Meeting, New Orleans, Louisiana, USA. https://agu.confex.com/agu/fm17/meetingapp.cgi/Paper/289480
- Taie Semiromi, M., & Koch, M. (2019). Analysis of spatio-temporal variability of surface–groundwater interactions in the Gharehsoo river basin, Iran, using a coupled SWAT-MODFLOW model. *Environmental Earth Sciences*, 78(6), 1–21. https://doi.org/10.1007/s12665-019-8206-3

Taniguchi, M., & Sharma, M. L. (1993). Determination of groundwater recharge using the change in soil temperature. *Journal of Hydrology*, 148(1), 219–229. https://doi.org/10.1016/0022-1694(93)90261-7

- Taviani, S., & Henriksen, H. J. (2015). The application of a groundwater/surface-water model to test the vulnerability of Bracciano Lake (near Rome, Italy) to climatic and water-use stresses. *Hydrogeology Journal*, 23(7), 1481–1498. https://doi.org/10.1007/s10040-015-1271-0
- Taylor, R. G., Scanlon, B., Döll, P., Rodell, M., van Beek, R., Wada, Y., et al. (2013). Ground water and climate change. *Nature Climate Change*, *3*(4), 322–329. https://doi.org/10.1038/Nclimate1744
- Terink, W., Immerzeel, W. W., & Droogers, P. (2013). Climate change projections of precipitation and reference evapotranspiration for the Middle East and Northern Africa until 2050. International journal of climatology, 33(14), 3055–3072. https://doi.org/10.1002/joc.3650
- Thampi, S. G., & Raneesh, K. Y. (2012). Impact of anticipated climate change on direct groundwater recharge in a humid tropical basin based on a simple conceptual model. *Hydrological Processes*, *26*(11), 1655–1671. https://doi.org/10.1002/hyp.8285
- Tilahun, K., & Merkel, B. J. (2009). Estimation of groundwater recharge using a GIS-based distributed water balance model in Dire Dawa, Ethiopia. *Hydrogeology Journal*, *17*(6), 1443–1457. https://doi.org/10.1007/s10040-009-0455-x
- Todd, D. K., & Mays, L. W. (2005). Groundwater Hydrology (3rd ed., pp. 656). New York, USA: John Wiley & Sons.
- van Engelenburg, J., Hueting, R., Rijpkema, S., Teuling, A. J., Uijlenhoet, R., & Ludwig, F. (2018). Impact of changes in groundwater extractions and climate change on groundwater-dependent ecosystems in a complex hydrogeological setting. *Water Resources Management*, 32(1), 259–272. https://doi.org/10.1007/s11269-017-1808-1
- VanderKwaak, J. E., & Loague, K. (2001). Hydrologic-response simulations for the R-5 catchment with a comprehensive physics-based model. Water Resources Research, 37(4), 999–1013. https://doi.org/10.1029/2000WR900272
- Varni, M., Comas, R., Weinzettel, P., & Dietrich, S. (2013). Application of the water table fluctuation method to characterize groundwater recharge in the Pampa plain, Argentina. *Hydrological Sciences Journal-Journal Des Sciences Hydrologiques*, 58(7), 1445–1455. https://doi. org/10.1080/02626667.2013.833663
- Vousoughi, F. D., Dinpashoh, Y., Aalami, M. T., & Jhajharia, D. (2013). Trend analysis of groundwater using non-parametric methods (case study: Ardabil plain). Stochastic Environmental Research and Risk Assessment, 27(2), 547–559. https://doi.org/10.1007/s00477-012-0599-4
- Wagener, T., & Gupta, H. V. (2005). Model identification for hydrological forecasting under uncertainty. Stochastic Environmental Research and Risk Assessment, 19(6), 378–387. https://doi.org/10.1007/s00477-005-0006-5
- Wilby, R. L., & Dawson, C. W. (2013). The Statistical DownScaling Model: Insights from one decade of application. International Journal of Climatology, 33(7), 1707–1719. https://doi.org/10.1002/joc.3544
- Wilby, R. L., Dawson, C. W., & Barrow, E. M. (2002). SDSM A decision support tool for the assessment of regional climate change impacts. Environmental and Modelling Software, 17, 145–157.
- Woldeamlak, S. T., Batelaan, O., & De Smedt, F. (2007). Effects of climate change on the groundwater system in the Grote-Nete catchment, Belgium. *Hydrogeology Journal*, 15(5), 891–901. https://doi.org/10.1007/s10040-006-0145-x
- Wu, Q. H., Wang, G. L., Zhang, W., Cui, H. D., & Zhang, W. (2016). Estimation of groundwater recharge using tracers and numerical modeling in the North China Plain. Water, 8(8). https://doi.org/10.3390/w8080353
- Yidana, S. M., Vakpo, E. K., Sakyi, P. A., Chegbeleh, L. P., & Akabzaa, T. M. (2019). Groundwater-lakewater interactions: An evaluation of the impacts of climate change and increased abstractions on groundwater contribution to the Volta Lake, Ghana. *Environmental Earth Sciences*, 78(3), 1–16. https://doi.org/10.1007/s12665-019-8076-8